The Legal Origins of Financial Development: Evidence from the Shanghai Concessions

Ross Levine, Chen Lin, Chicheng Ma and Yuchen Xu^{*}

November 2022

The critical challenge to assessing the legal origins view of comparative financial development is identifying exogenous changes in legal systems. We assemble new data on Shanghai's British and French concessions between 1845 and 1936. Two regime changes altered British and French legal jurisdiction over their respective concessions. By examining the changing application of different legal traditions to adjacent neighborhoods within the same city and controlling for military, economic, and political influences, we offer new evidence consistent with the legal origins view: the financial development advantage in the British concession widened after Western legal jurisdiction intensified and narrowed after it abated.

^{*} Levine: Haas School of Business, University of California, Berkeley, and the NBER, Berkeley, CA 94720-1900, (e-mail: rosslevine@berkeley.edu). Lin: Faculty of Business and Economics, The University of Hong Kong, Pokfulam Road, Hong Kong, (e-mail: chenlin1@hku.hk). Ma: Faculty of Business and Economics, The University of Hong Kong, Pokfulam Road, Hong Kong, (e-mail: macc@hku.hk). Xu: Peking University HSBC Business School, Nanshan District, Shenzhen, China, (e-mail: xuyuchen@phbs.pku.edu.cn). We thank Ying Bai, James Fenske, Peter Koudijs, Luigi Pascali, Hans-Joachim Voth, two anonymous referees, two associate editors, an expert reader, and the editor for helpful comments and suggestions.

The legal origins view of comparative financial development stresses that economies with a common law tradition more effectively protect private contracts and investors, enhancing the functioning of financial markets and institutions (La Porta et al. 1997, 1998). The view further asserts that legal traditions developed in Europe were spread to other countries through conquest and colonization. Researchers exploit differences in legal origin as exogenous determinants of legal system differences and financial development. Extensive research confirms the prediction of the legal origins view: countries with a common law tradition have more developed financial systems than those with a French civil law tradition (e.g., La Porta et al. 1997, 1998, 2002, 2006; Beck et al. 2000; Levine et al. 2000; Barth et al. 2004; Durnev and Kim 2005; Djankov et al. 2007, 2008; Morck et al. 2005; Berkowitz and Clay 2005, 2011; Brown et al. 2013; D'Acunto 2019).

However, as stressed in La Porta et al.'s (2008) literature review, the critical challenge to assessing the legal origins view is the degree to which one can interpret legal origin as an exogenous source of variation in the legal framework shaping financial development (e.g., Pistor 2009; Musacchio and Turner 2013). A country's legal tradition could proxy for culture (e.g., Stulz and Williamson 2003; Licht et al. 2005; Voigtländer and Voth 2012; Pascali (2016), Pierce and Snyder 2017; D'Acunto et al. 2019), social capital (e.g., Guiso et al. 2004), politics (e.g., Glaeser and Shleifer 2002; Rajan and Zingales 2003; Pagano and Volpin 2005; Perotti and von Thadden 2006; Roe 2006), or other factors omitted from the analyses. Aware of this overarching concern, researchers have controlled for many potential confounding factors to identify the impact of legal systems on the operation of financial markets (e.g., Beck et al. 2003a; Djankov et al. 2003; Fisman and Love 2004; Jappelli et al. 2005; Safavian and Sharma 2007; La Porta et al. 2006, 2008; Brown et al. 2013). Nevertheless, without randomly assigning legal systems to otherwise similar economies and then assessing the ramifications on financial development, the challenge of identifying the impact of legal origin on financial development remains.

In this paper, we exploit the unique history of Western concessions in China from 1845 through 1936 to assess the legal origins view of financial development. In Shanghai, China granted the British and French concessions over adjacent and very similar plots of land following the First Opium War. The lands were uncultivated and thus had no banks or financial markets before China conceded them to Britain and France. The British applied the common law in their concession, and the French concession used the French civil law. We evaluate the impact of applying the common and civil law on financial development within the same city. Besides the spatial discontinuity in legal traditions, another unique advantage of examining the Shanghai concessions is that there were two pivotal changes in legal jurisdiction over the concessions. The first change intensified the application of British common law and French civil law in their respective concessions. The second eliminated the application of Western legal traditions in their concessions. Thus, we evaluate two changes in the degree to which the British common and French civil laws shaped financial activities and developments in the respective concessions. Examining the comparative financial development ramifications of changing the application of different legal traditions on similar plots of land within the same city addresses several identification challenges associated with cross-country comparisons.²

To motivate and frame our econometric strategy, we briefly describe the two pivotal changes in legal jurisdiction over the Shanghai concessions, offering a fuller description in Section I below. From the founding of the British (1845) and French (1849) concessions until 1869, the respective consulates handled legal cases. During this period, the British and French administered their concessions, but Western powers had yet to operationalize their respective legal systems formally. Over time, more Chinese individuals and businesses moved into the concessions, increasing the number of "mixed" cases involving Chinese and foreigners. In 1869, there was the first pivotal change in the concessions' legal systems: The British and French established their own Mixed Courts to handle the full array of legal disputes, including mixed cases. The British Mixed Court used the common law, and the French Mixed Court applied the French civil law. There was a second regime change in the legal systems governing the concessions in 1926. The Provisional Rendition Agreement, signed in 1926, triggered a switch from the British common law and the French civil law to the Chinese civil law in the respective concessions. While maintaining administrative control, the Western powers returned legal jurisdiction to Chinese-run courts. These two regime changes in the legal systems governing the concessions define the critical breakpoints in the difference-in-differences methodology described below. Since British and French administrative influence over the concessions started to wane with the Second Sino-Japanese War, we end our analyses in 1936.

We manually collect data on financial development. Our primary measure of financial development is the number of banks (per square kilometer) in each concession and year. Although many studies of bank development use measures of credit as a share of gross domestic product (GDP) (e.g., King and Levine 1993), data on bank credit and concession GDP are unavailable. Thus, we focus on bank density. The bank density measure includes data on foreign banks, Chinese modern (joint-stock) banks, and Chinese traditional banks (qianzhuang). We compute the exact location of each

 $^{^{2}}$ Brown et al. (2017) examine the implications of imposing federal or state courts on different Native American reservations. We study the impact of imposing different legal systems on different areas of the same city.

bank by combining historical information on street addresses and mapping those addresses to GIS coordinates. We also compile annual information on the stock returns of all traded companies after 1866. Using these data, we assess how financial development changes following changes in legal jurisdiction over the Shanghai concessions.

To assess the impact of the British common law and the French civil law on comparative financial development, we begin by evaluating the first regime change in legal jurisdiction over the concessions using a difference-in-differences strategy. As noted above, creating the Mixed Courts in 1869 was a regime change, as the British and French formally implemented their legal systems in their respective concessions. The legal origins view predicts that financial development in the British concession should grow relative to that in the French concession after 1869. The difference-indifferences analyses include concession and year effects to help isolate the independent relationship between the application of the two legal traditions and comparative financial development. To further distinguish the "legal effect" associated with creating the Mixed Courts from other factors, we control for the "concession effect." Specifically, there is at least a 20-year period from the founding of the concessions until the 1869 creation of the Mixed Courts when Britain and France administered the concessions but did not have complete legal jurisdiction: there was a concession effect but not a legal effect during this period. By controlling for differences in comparative financial development during this pre-Mixed Courts period, we assess whether financial development rose comparatively faster in the British concession after the establishment of the Mixed Courts in 1869 above and beyond any differences triggered by the establishment and administration of two concessions.

Consistent with the legal origins view of comparative financial development, bank density in the British concession grows comparatively faster than that in the French concession after 1869. The British concession had about two more banks per km² than the French concession in the average year before the 1869 establishment of the Mixed Courts. This gap increased by 6 to 9 banks per km² (in the average year) after 1869 (i.e., between 1870 and 1925). Furthermore, when examining parallel trends, there is no sign of a change in comparative financial development before 1869. However, there is a notable and statistically significant break in the evolution of comparative financial development following the application of the different legal traditions in the concessions. We also address two potential concerns. First, empire-building factors might have shaped foreign bank entry into the concessions rather than legal system differences. Thus, we also conduct analyses on the subsample of Chinese banks. We find that Chinese banks drive the boost in comparative financial development in the British concession, not the entry of foreign banks. Second, although historical descriptions indicate that Britain and France established concessions on similar, adjacent lands, we control for potential differences by dividing the concessions into 200-by-200-meter grid cells. We repeat the analyses at this more granular level while including grid-cell fixed effects to account for within-concession variation. The results hold.

We extend these analyses to further address a critical challenge to drawing inferences about the impact of the British and French legal traditions on comparative financial development: omitted variables. Although the analyses include concession and year-fixed effects and control for the concession effect, time-varying omitted factors could trigger a break in comparative financial development around 1869. We condition on five concession characteristics and five indicators of Britain's and France's global military, economic, and political power. Regarding concession characteristics, we control for differences in (a) non-legal governance institutions in concessions, (b) population, (c) infrastructure expenditures, (d) international trading companies, and (e) the impact of violent conflict in China on the concessions. Regarding features of the British and French empires, we control for differences in (a) the number of concessions in other Chinese cities, (b) the number of naval ships, (c) the number of war victories around the world, (d) differences in the political systems of Britain and France, (e) the Gross Domestic Product (GDP) of Britain and France, and (f) stock market performance in Britain and France. Furthermore, we were concerned that the formation of the concessions could have set in motion different financial system dynamics not fully captured by concession and year-fixed effects or the array of timevarying control variables. Therefore, we also include the interaction between a British concession dummy variable and both linear and quadratic trends.

The results are robust to controlling for these additional covariates: we find a sharp increase in comparative financial development in the British concession following the establishment of the British and French Mixed Courts. These findings alleviate concerns that an omitted factor drives the break in comparative financial development in 1869. While other historical factors shaped comparative financial development, the findings suggest a robust, independent impact of applying the two legal systems on comparative financial development.

As additional strategies for reducing the potentially confounding effects of other historical events, we conduct the analyses over subperiods and after aggregating the data. First, we limit the sample period to the Qing dynasty (1840-1910) and thereby exclude the tumultuous change to Republican rule. Second, we restrict the sample period to the comparatively peaceful period between the Taiping Rebellion and the First Sino-Japanese war (1864-1894). The subsample analyses assess whether the results hold for narrower windows around the 1869 formation of the Mixed Courts. Third, we use data averaged over five-year periods. Using these data focuses on the longer-run relationship between legal traditions and comparative financial development. All results hold when implementing these additional strategies.

Next, we investigate the second regime change in legal jurisdiction over the concessions: the rendition of the Mixed Courts in 1926. Without changing the administration of the concessions, the rendition returned legal jurisdiction to Chineserun courts. The legal origins view predicts that since the British common law provides a comparative advantage over the French civil law in supporting financial activities, the rendition of the Mixed Courts will exert a comparatively strong adverse impact on financial development in the British concession. To test this prediction, we again employ a difference-in-differences approach. The first difference in these rendition analyses compares the British and French concessions. The second difference now compares the years before and after the 1926 rendition of the Mixed Courts to Chinese control. We again include the complete set of time-varying concession and empirical controls. Examining the 1926 rendition further reduces concerns about omitted variable bias, as omitted variables would have to account for the findings concerning both the 1869 and 1926 changes in legal systems.

The results from the rendition analyses are consistent with the legal origins view of comparative financial development, including extensions that examine bank performance and stock returns. The British concession experienced a much sharper decline in bank density following 1926. For this period, we obtained data on individual bank performance, as measured by return on assets (ROA). Consistent with the view that rendition had an especially adverse impact on banking in the British concession, we find a comparatively large drop in ROA among banks in the British concession. Evidence from changes in stock returns further supports the legal origins view. We examine a 24-month window around the August 1926 Provisional Rendition Agreement. We discover that stock returns in the concessions fall materially following the rendition of the Mixed Courts, and the fall is more prominent among firms in the British concession. Our joint findings that (a) the British financial development advantage surged after the 1869 application of the British common and French civil legal traditions in the respective concessions and (b) this gap shrank with the 1926 rendition of the Western legal systems to Chinese-run courts support the legal origins view.

As a final extension, we examine concessions in three cities outside of Shanghai: Guangzhou, Hankou, and Tianjin. We study 16 concessions granted to countries with a common law (Britain and the United States) or civil law tradition (Austro-Hungarian Empire, Belgium, France, Germany, Italy, Japan, and Russia). We analyze these three cities because each contains at least one common law and one civil law concession within its borders. Thus, we continue to compare financial development across concessions with different legal systems in the same city. There are limitations to examining these 16 concessions, however. These concessions did not experience the sharp intertemporal breakpoints in legal jurisdiction that characterize the Shanghai concessions. This limitation reduces the ability to distinguish the effects of legal tradition on comparative financial development from other features of the concessions that might shape financial development. Nevertheless, we provide these analyses to augment the examination of the Shanghai concessions. Consistent with the Shanghai results, we discover that financial development is materially greater in the common law concessions than in their civil law counterparts.

Our work relates to several lines of research. We contribute to the broad literature on how European colonization influenced comparative economic development by examining the impact of Western legal traditions on financial development in China (e.g., Acemoglu et al. (2001) and the review by Levine (2005)). We also add to a growing body of research on how the West shaped Chinese economic development during the 19th and 20th centuries (e.g., Fairbank 1978; Feuerwerker 1983; Kung 2021). Several recent studies explore how Western influences on international trade and the diffusion of knowledge following the First Opium War (1840-42) affected Chinese economic growth, firm dynamics, and domestic trade, e.g., Jia (2014), Bai and Kung (2015), and Keller et al. (2017), and Keller and Shiue (2021). Xin and Yan (2021)compare risk management by banks in the British and French concessions in the 1930s. They argue that the French concession has a more authoritarian governance structure than the British concession, creating a less stable environment and lower loan-todeposit ratios. More closely related to our work, Keller and Shiue (2021) examine the connections between Western legal institutions and interest rates across Chinese prefectures between 1820 and 1900. Using the existence of a Western consulate to gauge the presence of Western legal institutions, they find a negative relationship between having a consulate and prefecture interest rates and show that this consulaterate relationship is driven mainly by whether the prefecture has a British consulate. We focus on identifying the impact of legal origins on comparative financial development by (1) examining two adjacent plots of land in the same city, (2)evaluating the impact of two legal-specific changes (the establishment of the Mixed Courts in 1869 and the rendition of those courts in 1926) on comparative financial development in the two concessions, and (3) analyzing the response of foreign banks, Chinese modern banks, Chinese traditional banks, joint-stock companies, and stock returns to these legal-specific changes. We stress that (a) the British and French consulates administered their respective concessions from before the first until after the second legal-specific change and (b) the analyses condition on an array of concession and empire factors to isolate the impact of legal reforms on comparative financial development. Finally, we confirm that the results from the Shanghai concessions hold when examining other cities in China with multiple Western concessions.

We organize the remainder of the paper as follows. The next section provides a more detailed history of the Shanghai concessions. Section II describes the data. Section III presents the results analyzing changes in comparative financial development after Britain and France created the Mixed Courts. Section IV details the analyses and findings on changes in comparative financial development associated with the rendition of the courts to Chinese rule. Section V investigates the relationship between different legal systems and comparative financial development in cities outside of Shanghai. Section VI concludes.

I. Shanghai Concessions

In 1845, the British concession in Shanghai was formally established after Britain defeated China in the First Opium War (1839-1842) and demanded greater access to Chinese markets. Britain forced the Qing Dynasty to cede 830 acres of land in Shanghai. This "concession" was an enclave for British settlement, business, and trade, not a region of military occupation. In later years, China agreed to Britain's request to expand the concession as the number of foreign and Chinese settlers increased. The United States also received a Shanghai concession bordering the British concession. The British and American concessions merged in 1863, later named the Shanghai International Settlement. It reached a maximum size of 33,503 acres in 1899. Although more than ten countries had consulates within the Settlement, Britain dominated the administration of the International Settlement. For example, British citizens accounted for about 90% of the executive members of the concession's Municipal Council (Wang 1998). For this reason, we follow the literature and use the term "British concession" in referring to the International Settlement.

In 1849, the French also established a concession bordering the British concession in Shanghai. Specifically, France signed the Treaty of Huangpu with the Qing dynasty, obtaining the same trade and settlement privileges as Britain's concession. Fauvel (1899) notes that the first French Consul in Shanghai, Charles de Montigny, required that the concession be accessible by river and closer to Shanghai than the British concession. The Shanghai Prefect agreed, and the French concession was established on land between the British concession and Shanghai. The Suzhou River forms the border between the two concessions. The French concession also

experienced several expansion waves, from the original 164 acres to over 15,000 acres in 1914. Figure 1 illustrates the geographic expansion of the Shanghai concessions.

After 1936, British and French influence over the concessions waned appreciably with the Second Sino-Japanese War (1937-1945). As a result, we end our analyses of the Shanghai concessions in 1936. Britain and France formally returned the concessions to the Chinese government in 1943.

A. The British and French legal systems in the concessions

Britain and France transplanted their common and civil law legal systems into their respective Shanghai concessions. During the initial decades, the British and French consulates handled disputes involving their citizens. As more Chinese moved into the concessions and economic activity boomed, the British and French consulates struggled to address the burgeoning array of "mixed" cases that involved financial and commercial disputes between Chinese and foreign individuals, firms, and banks. In 1863, officials began negotiating an arrangement to replace these consular courts with legal institutions that could better serve the community's needs (Hudson 1927).

In 1869, a new charter was finalized, and the British and French concessions each established their own Mixed Courts (*Huishen Gongxie*) to handle cases involving Chinese and foreign plaintiffs and defendants. The 1869 Statutes of the Mixed Courts in Shanghai set clear, formal jurisdictional boundaries between the Mixed Court of the International Settlement (British Mixed Court) and the Mixed Court of the French Settlement. The Statutes specified which of the Mixed Courts would adjudicate cases involving people and businesses within the concessions, as detailed in Jernigan (1905, pp. 203-4). There were also detailed rules concerning legal jurisdiction when disputes included people, businesses, or properties outside the concessions. The British and French courts handled disputes involving concession residents (foreigners or Chinese) or their assets (Lee, 1990; Hou, 2017b). For instance, if a bank in the British concession sued a client in another Chinese district, the British Mixed Court would hear the case. When the plaintiffs were creditor banks in other Chinese cities, and the defendants were residents of or businesses in a concession, the corresponding Mixed Court in Shanghai would try the case.

Furthermore, the Mixed Courts worked with Chinese authorities to enforce the law beyond the concessions. For example, the Chinese authorities would apprehend defendants outside of the concessions for the Mixed Courts (Lee, 1990). During the initial decades of the Mixed Courts, Chinese and foreign judges jointly heard cases involving Chinese citizens and foreign residents (see an illustration in Internet Appendix Figure A1), but foreign judges dominated the courts. As noted by the U.S. State Department (1880, p. 158), Chinese officials mainly assisted in executing decisions. Research by Hudson (1927), Yang (2006), and Cai (2013) notes that the Chinese officials participating in the Mixed Courts (a) tended to have a very low rank within the Chinese hierarchy, (b) were chosen by foreign officials to serve in the Mixed Courts, and (c) had little to no influence over decisions. By 1911, the Mixed Courts had jurisdiction over civil cases, including cases involving only Chinese citizens within the concessions, as well as the appointment, supervision, and payment of Chinese judges, control over prisons, the execution of arrests, etc. (Hudson 1927; Shanghai Shehui Kexueyuan 1981).

Around 1926, there was another "regime change" in the legal system governing the concessions. Due to rising nationalism after the First World War, China began negotiating a plan of rendition whereby the Mixed Courts would surrender legal jurisdiction to the Chinese authorities (Hudson 1927). These negotiations started in 1916 but failed to reach an agreement. In 1925, violent demonstrations in the concessions over foreign "imperialism" reenergized negotiations about rendition. The "Provisional Rendition Agreement" was signed on August 31, 1926, and the formal rendition of the British Mixed Courts to Chinese-run courts occurred on January 1, 1927. This was a significant change as it involved a switch from British common law to Chinese civil law. The French Mixed Court was also materially altered in 1927, as Chinese cases were assigned to Chinese judges and lawyers. Then, in 1931, Chinese courts assumed complete legal jurisdiction over the French concession.

A.1. Common law in the British concession

The historical record of lawsuits demonstrates that the British concession followed a common law tradition. The courts decided cases based on a mixture of statutory law, precedent, and the application of those laws and precedents to emerging cases within the concession.³ In the concession, the British common laws applied to foreign residents and local Chinese. With the accumulation of court cases in the concession, the Court gradually invoked local cases as precedents in deciding new cases, which is a typical manifestation and evolution of the British common law outside of Britain (e.g., Levine 2005). For example, in a 1911 case, an American publishing house, Messrs, Ginn & Co., sued the Commercial Press of China for copyright infringement. The lawyer cited two verdicts from cases in 1896 and 1907 within the British concession

³ We confirmed the operation of the British common law in the Shanghai concession by reviewing commercial and financial lawsuits regarding property rights, financial instrument transactions, bankruptcy, creditors' residual rights after the liquidation of a company, and shareholders' rights and shares in joint ventures. It is worth noting that well-developed law libraries emerged in the Municipal Council of the British concession, which facilitated the law practitioners of the day and our research.

in which Chinese businesspeople were convicted of pirating foreign books. The American publishing house won the case, and Commercial Press had to recompense Messrs, Ginn & Co. This case is discussed in a series of newspaper reports in the *North China Herald*, April 8, April 10, and October 7, 1911.

Another noteworthy feature of the common law is its ability to adapt and consider local circumstances and customs in addressing specific cases, as discussed in Beck, Demirgüç-Kunt, and Levine (2003b) and La Porta, Lopez-de-Silanes, and Shleifer (2008). Evidence of this feature exists in the British concession, as legal practitioners regularly refer to Chinese customs when addressing disputes. Thus, without violating common law principles, the law could sometimes evolve efficiently in the British concession to resolve disputes while respecting local customs. For example, in 1876, Sassoon & Co., Ld., E.D. sued Wang, the owner of a Chinese company, in the mixed court for the overdue payment of a promissory note. Wang argued that the promissory note with 500 tael principal was transferred to the Chinese merchant Guo to purchase opium. However, Guo's opium business went bankrupt, which led to the suspension of the repayment of the note. The verdict was that since there is a stamp of the Chinese company on the promissory note, Wang must repay the note to Sassoon & Co. The judgment was based on the Chinese convention that the validity of notes depends on an official Chinese seal. If there is a default on the payment, the issuer with a seal on the promissory note is financially responsible.⁴

A.2. Civil law in the French concession

Consistent with the French civil law tradition, the French Mixed Court decided cases based on the statutes passed in France, not on precedent or local custom. The Court's archives show that judges resolved commercial cases based on strict applications of French commercial codes.⁵ For example, in 1910, a French merchant sued a Chinese broker who traded gold futures for him, charging that the speculative behavior of the Chinese broker resulted in the French merchant losing more than 10,000 silver taels. Citing *Section 1965 of the French Civil Code* on the recovery of debts, the French court rejected the French merchant's argument, reasoning that the contract did not specify precise limits on trading by the Chinese broker. So, there was no basis for classifying the broker's trade as illegal (Hou (2017) cites this case "Séance du 27 November 1907 S. Somekh contre Hi Kai Song" [no. 635PO/C/371] from the Diplomatic Archives reserved in Nantes of France). Also consistent with the French

⁴ Cai (2013) cites this case, which is discussed in the article "Case on opium business promissory note (Zhipiao Fayin)" from the Chinese daily newspaper, Shun Pao, on January 22, 1876.

 $^{^5}$ The cases are reserved in the Diplomatic Archives in Nantes of France.

civil law tradition, judges in the French concession exerted power over investigating cases, calling and questioning witnesses, etc. Accordingly, lawyers in the French concession—as in France—mainly dealt with drafting and submitting documents to the court and replying to judges' queries (Hou 2017). Finally, we reviewed all civil cases in the two concessions. Relative to cases in the British concession, more than three times as many cases in the French courts cite statutory codes, and less than half cite earlier cases, consistent with the comparative law literature's description of the two legal traditions.

B. Financial development

The concessions were established on relatively undeveloped land, with few prior inhabitants and no pre-existing financial institutions or markets. For example, Montalto de Jesus's (1919, pp. 27-29) history of Shanghai describes the land ceded for the British concession as uncultivated, with several creeks and a few small hamlets. Likewise, French settlers described the initial condition of their concession as "primitive, covered with houses in poor condition, graves, and rubbish dumps" (Fauvel, 1899. p.17).⁶

By the late 19th century, however, Shanghai had become the most prominent financial hub in Asia, and the British concession was the financial center of Shanghai. For example, two years after the foundation of the British concession, the English Oriental Bank opened its first branch in China within the British concession. Foreign banks steadily entered the British concession, reaching 30 by 1925. In contrast, the total number of foreign banks in the French concession reached only four in the same period. These foreign banks dominated the foreign exchange market and foreign trade financing.

Over time foreign banks collaborated with Chinese traditional banks called qianzhuang ("money house") to provide a broader range of financial services to local businesses. The qianzhuang were generally small sole proprietorships or partnerships in which the founders were from wealthy merchant families. The qianzhuang blossomed in Shanghai after the mid-19th century. Their activities included providing local currency exchanges for the varieties of metallic currencies circulating in China, noteissuing, and giving credit to firms and households. While foreign banks had abundant capital, qianzhuang had local expertise. The two financial intermediaries formed fruitful business relations in the Shanghai concessions. Indeed, many qianzhuang moved their headquarters into the concessions. The qianzhuang even relocated their

⁶ Furthermore, there is no evidence that the British concession enjoyed a superior location. Indeed, as noted above, the French strategically chose their concession to give it an advantage over the British.

"Shanghai Monetary Association" (Shanghai qianye gonghui) from the Shanghai old city to the British concession in 1917.

According to the legal origins view, the British legal system offered advantages over the French civil law in accommodating the growth of Chinese traditional banks. Before the concessions, Chinese traditional banks developed their own business rules, enforced by their Association. The legal origins view stresses that the British common law tends to adapt to and incorporate local circumstances in enforcing contracts and deciding cases to a much greater degree than the French civil law (e.g., Beck, Demirgüç-Kunt, and Levine 2003b; Levine 2005; Gennaioli and Shleifer 2007; La Porta, Lopez-de-Silanes, and Shleifer 2008). As a result, the British concession would have a comparative advantage in fostering the development of Chinese traditional banks and facilitating collaboration between qianzhuang and British foreign banks. Indeed, British Mixed Court verdicts involving Chinese traditional bankers often cited the rules and customs of the Association of Chinese traditional banks (Du 2006), as shown in the archives of the British Mixed Court.⁷

In 1897, the Imperial Bank of China—the first "modern" Chinese bank—opened in the Shanghai British concession. Unlike the qianzhuang, the Imperial Bank of China was a limited liability, joint-stock company with regular shareholder meetings, a separate board of directors, and financial and accounting disclosure statements based on international standards. Many such modern Chinese banks soon emerged. By 1925, the number of Chinese banks in the British concession reached 194, more than five times that of foreign banks (30). However, the number of Chinese banks in the French concession grew much less rapidly, reaching 29 banks in 1925. Banks in the Shanghai concessions, especially those in the British concession, established branches and made loans outside the concessions. During the 1869-1926 period, 32% of Chinese banks in the British concession had branches outside the Shanghai concessions, whereas only 0.07% of banks in the French concession had such branches.

Besides modern banks, nonfinancial Chinese stock companies with limited liability and traded shares also emerged and flourished in the Shanghai concessions (Wang 1965). Before the concessions, there were no limited liability, joint-stock companies in China. Investors could purchase stocks directly from the company or through qualified brokers (Liu 2004). An active secondary market emerged in the 1860s. The most influential newspaper in Shanghai, *The North-China Herald*, published daily stock prices in "Shares and Stocks" (see Internet Appendix Figure A2). During the early years, stocks were traded mainly through foreign banks and foreign stock trading

⁷ For example, the British Mixed Court in 1918 decided that a promissory note issued by a trade company, Hongyu Hao, was invalid because the company was the subject of fraud. However, British Mixed Court reversed this decision after the Association noted that their rules specify that all certifiable promissory notes are redeemable regardless of fraud. There are many such cases discussed in Du (2006).

companies. The Shanghai Share Broker's Association was founded in 1891, giving rise to The Shanghai Stock Exchange in 1903. Most publicly-traded companies were in the British concession. For example, by 1925, 74 publicly-traded companies operated in the British concession and four in the French concession.

II. Data and Textual Analyses of Cases

This section first defines basic data from the British and French Shanghai concessions, including the unit of analysis and data on banks and stock prices. The section then provides textual analyses of court cases to illustrate differences in the operation of the British and French legal systems in the concessions. Below, when conducting robustness tests, we define additional control variables and discuss the data and details of the non-Shanghai concessions.

A. Unit of geographic analysis

We compare the development of finance in the British and French concessions from the establishment of the British concession in 1845 until the eve of the Second Sino-Japanese War in 1936, which reduced Western dominance over the concessions. We define the geographic boundaries of the concessions as those lands that were part of the concessions before 1861, which we call the "old concessions." The results, however, are robust to using the post-1861 geographic boundaries of the concessions. Figure 1 provides a map of the British and French concessions in Shanghai. As noted above, the British and French established concessions on similar, barren lands. Nevertheless, we account for potential variation in land quality, infrastructure, and other locationspecific effects by dividing the concessions into 100 200m × 200m grid cells and including grid cell fixed effects in the analyses below. We exclude grid cells less than 0.001 km², which leaves 100 grid cells. Internet Appendix Table AI provides descriptive statistics of the main variables.

B. Banks and stock returns

We calculate the number of bank headquarters and branches in a concession or grid cell each year, as banks were the primary financial intermediaries in the late 19th and early 20th centuries. We do not have information on each bank's assets (or liabilities). Shanghai banks included foreign, Chinese modern, and Chinese traditional banks (i.e., the qianzhuang described above). Banks took deposits, made loans, and provided transaction services. Only foreign and Chinese modern banks had branches, as the qianzhuang had only one office (headquarters). For cases where a foreign bank headquartered outside of China has branches in a concession, we count the branch as a bank in the concession. When examining grid cells, we compute the number of headquarters and branches within each grid cell.

We manually collected information on each bank's address and years of operation from the *Shanghai Financial Gazetteer* and other archives and newspapers (Internet Appendix B1 lists the data sources). We then map the addresses to the geographic information system (GIS) coordinates and identify the banks within concessions and grid cells. There were 67 foreign banks, 335 Chinese modern banks, and 264 qianzhuang in the two Shanghai concessions from 1846 through 1936.⁸ Figure 2 depicts the distribution of banks across the grid cells of the two concessions, using the boundaries of the "old concessions." The figure provides the distributions of banks in 1868, which is just before the creation of the Mixed Courts, and in 1925, which is just before the rendition of the Mixed Courts to the Chinese courts. As shown, there is much more rapid financial development in the British concession than in the French concession following the creation of the Mixed Courts, which established the British and French legal systems in their respective concessions.

We also collected data on banks' return on assets (ROA), assets, leverage (loan to asset ratio), and age from a 10-year survey by the Research Department, Bank of China (1933) covering the 1921 to 1931 period. Although we could only obtain data on 26 banks, we use them to shed additional evidence on the impact of the rendition of the Mixed Courts on comparative bank performance between the two concessions.

Finally, we assembled data on the stock returns of individual firms from the *North-China Herald* and *China Press*. The *Herald* was founded in 1850. It began to publish stock prices in 1866 and the number of shares in 1894. The *China Press* published stock prices in Shanghai between 1911 and 1949.⁹ We manually identified the exact location of 205 stock company's location (coordinates) based on *The Desk Hong List*, a bi-annual survey of all Shanghai firms edited by the *North-China Herald* between 1872 and 1941. The survey provides detailed information on each company's name, location, business scope, and other characteristics. Then we mapped the companies to the grid cells of the two concessions.

C. Court cases: Aggregate data and textual analyses of individual cases

 $^{^8}$ These numbers omit the 43 Chinese modern bank branches and 86 qianzhuang, for which we could not identify the precise location.

⁹ These stock return data do not include dividends, as there are no firm-level data on dividends during this period.

We collected data on civil cases to document differences in the functioning of the British common and French civil law systems in the concessions. We first collected summary information on all civil cases from the Municipal Annual Reports of each concession (1908-1926). Most civil cases in both concessions involve contractual disputes related to business contracts, debts, etc. The British court tried 1,132 civil cases per year, about 2.6 times the number of cases in the French court. To assess differences in the functioning of the two courts, we digitized the texts of the verdicts for 460 commercial cases, 315 from the British concession and 145 from the French concession. For the British court, we found 315 commercial cases from the total of 6,201 cases reported in the Supreme Court & Consular Gazette of the North-China Herald from 1870 to 1926. For the French court, we photocopied all court cases from the Diplomatic Archives (1903-1926) reserved in Nantes of France, which provides the most extensive collection of the original verdicts from the French court in the Shanghai concession.

We then conduct textual analyses of the verdicts over the 1870-1926 period. We differentiate by three sets of keywords to gauge differences in how the Britishand French legal systems addressed commercial disputes in the concessions. First, we use keywords associated with custom and convention, as the legal origins view holds that the British common law is more likely to use local customs and conventions in deciding cases than the French civil law. We then calculate the ratio of the number of keywords associated with "custom and convention" to the total number of words in the verdict and multiply by 1,000 to compute the millesimal. Second, we use keywords associated with consult and opinion, as the legal origins view stresses that the British common tends to refer to past opinions and consultations more generally in interpreting the law. In contrast, the French civil law focuses more on statutory laws written in France when deciding cases. We compute a similar ratio of the number of keywords associated with "opinions and consultations" to the total number of words. Third, as a falsification test, we examine the ratio of keywords associated with default, infringement, bankruptcy, etc., to the total words in the verdict. We examine these keywords associated with contractual disputes as a falsification test because the legal origins view does not suggest that there will be proportionally more such disputes in British or French concession. Appendix Table AII provides the keywords from the verdicts. Finally, we test whether the incidence of these three sets of keywords differs between the British and French Mixed Courts.

Table I reports the findings for all cases and the subset of bank-related cases. Compared to cases from the French concession, cases in the British concession are more likely to use words such as custom or convention, consistent with the view that the common law is more likely to consider local customs and general convention in resolving disputes. Verdicts in the British concession more frequently use words such as opinion and consult, consistent with the view that the common law considers various factors to resolve cases efficiently, not just the statutory law. However, there is not a higher proportion of cases using words like "default" These findings consistent with the comparative law literature's descriptions of the differences between the British common and French civil law. The British common law relies more on local customs and conventions and past opinions, while the French civil law focuses more on statutory laws developed in France.

III. Financial Divergence between the Concessions: 1840-1925

A. Methodological overview

The critical challenge to assessing the impact of the British common law and the French civil law on comparative financial development in their respective Shanghai concessions is isolating the influences of legal system differences from other potential factors. As noted above, the British and French concessions were on neighboring lands of similar quality that had similar pre-concession levels of economic development. Britain and France, however, were different in other potentially important respects besides their legal systems. For example, Britain had a larger economy and a more extensive colonial empire. As a result, commerce and finance might have evolved differently in the British and French concessions for reasons having little to do with legal systems.

In this section, we use a series of analytical methods to isolate the independent role of legal system differences in shaping comparative financial development from other differences between Britain and France. First, we begin with a simple framework in which we explore differences in financial development between the two concessions following the formation of the Mixed Courts in 1869 to assess the "legal effect." We control for the "concession effect" associated with the formation of the concessions in 1845 and year and concession (or grid-cell) fixed effects to isolate the legal effect on comparative financial development. Second, we test the robustness of the findings by (a) using different subperiods to reduce the impact of some major historical events on the analyses and (b) aggregating the data from annual to five-year periods to reduce the impact of annual fluctuations on the findings. Third, we assess whether there is a break in the evolution of comparative financial development around the formation of the Mixed Courts in 1869. Finally, we extend these analyses by conditioning on an array of time-varying concession, China, and empire traits, including measures of the comparative economic, military, and political strengths of Britain and France. We begin with the following difference-in-differences equation to assess the impact of the two concessions and their legal institutions on comparative financial development:

$$Y_{it} = eta_1 imes British_i imes Post1845 + eta_2 imes British_i imes Post1869 + \gamma_i + \lambda_t + arepsilon_{it}.$$
 (1)

 Y_{it} denotes the number of banks (per km²) in a concession (or grid cell) *i*, in year *t*. British_i is a dummy variable that equals one for the British concession (including all grid cells within the British concession when conducting the analyses at the grid-cell level) and zero for the French concession. Post1845 is a dummy variable that equals one after 1845 and zero before then. We also conduct the analyses at the grid-cell level because we were concerned that the British concession might have some especially appealing areas for financial development, perhaps due to land quality, location, or infrastructure. These within concession factors might account for comparative financial development in the British concession and not legal system differences.¹⁰ Given persistence in the dependent variable, we use a Generalized Least Squares (GLS) estimator designed for panel data to account for potential autocorrelation throughout our analyses.¹¹

The first interaction term, $British_i \times Post1845$, captures the concession effect, i.e., whether more banks started operating in the British concession relative to the French concession after 1845. Since there were no banks in the concessions before 1845, the estimated coefficient on $British_i \times Post1845$ provides an estimate of the average annual difference in bank density between the British and French concessions following the creation of the concessions. γ_i is a concession (or grid-cell) fixed effect that captures all time-invariant characteristics of the geographic location. λ_t is a year fixed effect and captures the common annual shock shaping financial development in both the British and French concessions.

The second interaction term, $British_i \times Post1869$, captures the legal effect, i.e., whether more banks were opened in the British concession relative to the French concession after their respective legal systems became dominant in the concessions with the creation of the Mixed Courts in 1869. By controlling for the concession effect,

¹⁰ Since there are only two concessions, we do not adjust standard errors for clustering at the concession level when running concession level regressions. When running grid-cell level regressions, we adjust the standard errors for clustering at the grid-cell level.

¹¹ The GLS estimation considers the autocorrelation between residuals of the estimates by assuming the residuals follows an AR(1) process and assigning them an autocorrelation parameter that is calculated by $\epsilon^{\prime} \epsilon$ (t-1)/ $\epsilon^{\prime} \epsilon$, where ϵ is the vector of residuals.

British_i × Post1845, the estimated coefficient on the legal effect, British_i × Post1869, provides information on whether bank density rose comparatively faster in the British concession after the establishment of the Mixed Courts in 1869 beyond any differences in the evolution of bank density triggered by the establishment of the two concessions in 1845. Thus, $\beta_2 > 0$ would suggest that bank density rose in the British concession relative to the French concession beyond the formation of the concessions in 1845 after establishing the different legal systems in the concessions.

As stressed above, the establishment of the Mixed Courts represents a distinct break in the concessions. The creation of these courts deepened and broadened the jurisdiction of the British and French legal systems in their respective concessions. Thus, the "legal origins" view of LLSV (1998) suggests that (1) financial development in the British concession should be comparatively stronger than that in the French concession after 1869 relative to differences in financial development in the two concessions before the establishment of the Mixed Courts and (2) the impact of the legal systems on comparative financial development should hold above and beyond any effects of the British and French global military, economic, and administrative characteristics on comparative financial development. We start our examination of the legal origins view by using data from 1840 through 1925, where the 1840-1845 period is the pre-concession period.

As shown in Table II, the results indicate that bank density in the British concession was markedly greater than that in the French concession after establishing the Mixed Courts in 1869. Consistent with the view that the 1869 imposition of the different legal systems—and not simply the creation of separate British and French concessions—altered comparative financial development, the results on $British_i \times$ Post1869 hold when conditioning on the concession effect (British_i × Post1845). Based on the coefficient estimates from column (1), the British concession had about 2 more banks per km^2 on average than the French concession before establishing the Mixed Courts in 1869, and this gap increased by about 8 banks per $\rm km^2$ after 1869. In column (2), we re-do these analyses for the subsample of Chinese banks to net out the comparative number of foreign banks establishing affiliates in the concessions. The estimates indicate a comparatively sharp increase in the density of Chinese banks in the British concession, with the gap increasing by nearly 7 banks per $\rm km^2$ after the establishment of the Mixed Courts. By comparing columns (1) and (2), the estimates suggest that the legal effect is primarily composed of a large increase in Chinese banks in the British concession following the stronger application of the British common and French civil laws triggered by the establishment of the Mixed Courts. Furthermore, as shown in columns (3) and (4), these results hold for the full sample of banks and the subsample of Chinese banks when conducting the analyses at the grid-cell level.¹²

We next extend these analyses to address concerns that political, military and economic disruptions differentially shaped financial development in the two concessions by conducting the analyses over sub-periods and using lower frequency data. As stressed above, our analyses focus on comparative financial development. Thus, while political, military, and economic upheavals may have shaped financial and economic activity in the concessions, our concern is whether such disruptions differentially affected financial development in the concessions. In one set of subsample analyses, we limit the sample period to the Qing dynasty (1840-1910), which excludes the tumultuous change to Republican rule starting with the 1911 revolution. In another set of subsample analyses, we restrict the investigation to the comparatively peaceful period in China between the Taiping Rebellion and the First Sino-Japanese war (1864-1894), a period that also excludes the Boxer Rebellion (1899-1901). Both of these subperiods exclude World War I. By examining these subsamples, we test whether the results hold for narrower windows around the 1869 formation of the Mixed Courts and for periods that exclude several major events.¹³ As shown in Panels A and B of Table III, all of the results hold during these two subsamples. In a second approach, we use data averaged over five-year periods to abstract from higher-frequency relationships that could confound our ability to isolate the long-run relationship between legal traditions and comparative financial development. As shown in Panel C of Table III, the results hold when using these aggregated data, confirming the earlier results with annual data. In the next two subsections, we employ different strategies for identifying the impact of legal traditions on comparative financial development.

¹² The results hold when we normalize the number of banks by population size or take the natural logarithm of bank density at the concession level. Consistent with the findings on bank density, we discover that stock market development is comparatively strong in the British concession following the establishment of the Mixed Courts. To measure stock market development, we use the density of stock companies at the concession and grid-cell levels. The sample period begins in 1866, as there were no stock companies in Shanghai before then. This leaves only three years of data before the creation of the Mixed Courts. Therefore, we view the results on stock companies around 1869 as suggestive (not reported).

¹³ Another potential confounding event was the 1870 Tianjin Massacre in 1870, in which about 60 French Catholic clergies were killed by Chinese locals. While this incident led to tensions between China and France, it is unclear how it influenced the French and British concessions in Shanghai. On the one hand, the tensions might have discouraged Chinese merchants from investing in the French concession. On the other hand, France demonstrated its influence in China, as the Qing emperor punished officials and others engaged in the massacre. France's demonstrated power might have attracted Chinese merchants into the French concession. We have not found evidence that the power and prestige of France fell in China after the massacre.

C. Dynamics: Comparative financial development and the Mixed Courts

To provide additional evidence on whether there is a break in the comparative evolution of bank density in the British and French concessions after the 1869 formation of the Mixed Courts, we estimate a dynamic version of equation (1). Specifically, we run the following regression at the grid cell level with 1869 as the reference year:

$$Y_{it} = \sum_{t=1865}^{1880} \beta_t \times British_i \times \theta_t + \gamma_i + \lambda_t + \varepsilon_{it}, \qquad (2)$$

where θ_t equals one in year t and zero otherwise, and the other variables are defined above. To limit the effects of heterogeneity over longer periods and focus on a narrower period around the 1869 establishment of the Mixed Courts, we conduct this examination over the window from 1865 through 1880, i.e., five years before the establishment of the Mixed Courts ten years after that. Note, however, that if we conduct the analyses over a longer period, the same results emerge. We then plot the estimated values of β_t and the 95% confidence intervals in Figure 3. If the establishment of the Mixed Courts is associated with a positive change in the comparative evolution of financial development in the British concession, we should see an increase in β_t after 1869. We conduct the analyses for the full sample of banks (Panel A) and for the subsample of Chinese banks (Panel B).

Before the formation of the Mixed Courts, the gap in British-French bank density is close to zero and does not exhibit a significant trend, as shown in Figure 3. There is a notable change after 1869. Consistent with the legal origins view of comparative financial development, the financial development advantage in the British concession starts to grow after the formation of the Mixed Courts. Consistent with the view that it takes time for legal systems to shape financial development, Figure 3 illustrates a growing gap in British-French financial development.

D. Mixed Courts and bank development: Additional controls

In this section, we conduct additional analyses to enhance identification. The findings in Tables II and III and Figure 3 suggest that bank density in the British concession rose appreciably relative to that in the French concession after 1869 while controlling for the pure concession effect, concession (and grid-cell) fixed effects, and year fixed effects. We now control for an array of time-varying factors to reduce concerns that an omitted variable accounts for the surge in comparative financial development starting in 1869. Given the difference-in-differences specification and control variables, the omitted factors would have to be time-varying and induce a break in comparative financial development around the time of the formation of the Mixed Courts to confound our analyses. Thus, the omitted variables must not just affect financial development; they must differentially shape financial development around 1869 to bias the findings on the legal effect. Data on the concession-level controls that we now add to the analyses start only after the formation of the concessions, so this section's analyses begin in 1846.

We include ten additional time-varying controls. Five control for concession characteristics, and five control for differences in the British and French empires' military, economic, and political power. We first describe the data and then present the results. For brevity, we provide results for the subset of Chinese banks, but all results hold for the full sample of banks as shown in Appendix Table AIII. The summary statistics are reported in the Internet Appendix Table AI.

D.1. Shanghai concession controls

First, Britain and France established different governance institutions in Shanghai, and it could be these governance systems and not comparative legal institutions that shaped financial development (e.g., Xin and Yan 2021). Although we control for concession fixed effects, we extend these analyses by controlling for the dates when the British and French formally established different governance systems in their respective concessions. Historians highlight two substantial changes in the governance of the concessions: the British establishing their Municipal Council in 1854 and the French creating their own Municipal Council in 1862. We test whether the creation of these different governance systems accounts for the divergence in bank density in the concessions. To do this, we create the variable *Municipal Councils*, which equals one after 1854 for British concession observations, equals one after 1862 for French concession observations, and equals zero otherwise.¹⁴

Second, despite starting from similar economic conditions, factors besides legal system differences may have shaped comparative economic development and hence comparative bank development in the concessions. To address this concern, we control for population per square kilometer (*Population density (lag)*) to proxy for economic development since there are no data on production in the concessions. We use lagged population density at the concession-year level to assess the robustness of our results while recognizing that these population data are subject to several challenges (Zou,

¹⁴ Another possibility is that the British concession had lower tax rates, fostering greater economic and financial development. However, we find that this was not the case. As Wang (1995) shows, the tax rates on land and property taxes, where land and property taxes accounted for about 80 percent of fiscal revenues in the concessions, are almost identical in the two concessions.

1980; Henriot et al., 2018).¹⁵ Thus, if economic, political, or military events encouraged more people to move into the British concession to engage in commercial transactions, spurring financial development, including *Population density (lag)* will help in identifying the independent relationship between the legal effect and comparative financial development from these other influences.

Third, we were concerned that Britain might have spent more on infrastructure development around the formation of the Mixed Courts, which might have independently spurred economic and financial development. We collected data on annual public spending (measured in 1,000 silver Taels) on infrastructure and government services in the two concessions from their Annual Municipal Reports (Shanghai Municipal Council, 1863-1936; Conseil d'administraction municipal de la concession francaise a Shanghai, 1868-1936). We then control for *Public expenditures* to assess the independent relationship between the implementation of distinct legal systems in 1869 and the subsequent evolution of comparative financial development.

Fourth, the comparative strength of Britain's global military and economic empire might have encouraged relatively more economic activity and trade in its concession. To control for this potential confounding influence, we include *Trade company density*, which equals the number of trade companies per square kilometer at the concession level in the previous year, in the regressions. We obtained this data from the *Desk Hong List* edited by the *North-China Herald* between 1872 and 1941.

Fifth, we control for the possibility that violent conflicts within China differentially influenced financial development in the concessions. We construct the variable *Wars in China*, which equals the number of civil wars and foreign wars in China in the previous year. We then include the interaction term *British*Wars in China* to control for the possibility that conflicts within China differentially shape comparative financial development. We obtain these data from Chen (1992).

D.2. Empire controls

Differences between Britain and France in their military, economic, and political power may have shaped comparative financial development in the Shanghai

¹⁵ From 1865, both concessions conducted population census surveys every five or ten years. Annual population data start in 1900 in the British concession and the late 1920s in the French concession. Assuming a linear growth rate of population, we interpolated the annual population for the missing years and coded the population as zero in 1845. Another challenge to working with population data is that we do not have separate time series data on the geographic areas defined as the old concessions (our sample areas). Instead, we use the whole population of the British and French concessions, as the districts of the old concessions were the most populous. We make this assertion based on the data illustrated in Internet Appendix Figure A3 for the population distribution at the district level within the concessions in 1935, which is the only year with district-level population data across concessions.

concessions. Omitting such influences could confound our ability to draw sharp inferences about the influence of legal traditions on comparative financial development if those characteristics differentially shaped financial development in the concessions and induced a break in the evolution of comparative financial development around 1869. Thus, besides controlling for the concession-specific factors discussed above, we now also include six "empire controls" to condition out broader features of Britain's and France's global power and influence within China.

First, to the extent that Britain had greater economic influence and connections throughout China, the British Shanghai concession may have experienced comparatively stronger financial development than the French Shanghai concession. Although our earlier analyses controlled for the concession effect (*British*Post 1845*) and concession fixed effects, we control for the number of British and French concessions in China (*Concessions in China*) based on Yan's (1955) statistics.¹⁶

The second and third controls capture the global military power of the empires. Naval ships equals the number of naval ships in the British and French navies in the previous year. We obtain these data from Colledge and Warlow (2010), Demerliac (2013, 2021), Chesneau (1979, 1980), and Gardiner (1985). War victories equals the number of military victories involving Britain or France in the previous year, which may influence the relative power and reputation of Britain and France.¹⁷ These data are from Clodfelter (2002).

Fourth, differences in the political systems of Britain and France may also have shaped financial development in the concessions, and it might be these political factors, not legal system differences, that account for comparative financial development in the Shanghai concession. For example, data from the Polity IV database indicate that democracy grew in Britain during the sample period. In contrast, France remained comparatively authoritarian. If democracy fosters greater investor confidence, then

¹⁶ Hong Kong was an important British settlement, raising concerns that financial development in Hong Kong drove financial development in its Shanghai concession. However, the historical evidence suggests that this is very unlikely. Before the mid-20th century, Hong Kong was much less financially developed than Shanghai. For example, Hong Kong had 10 modern Chinese banks in the 1930s, while Shanghai had 218. Moreover, of the 10 modern Chinese banks in Hong Kong, eight were first established in Shanghai. Then, after many years, they opened affiliates in Hong Kong. None of the 218 modern Chinese banks in Shanghai opened first in Hong Kong. The historical evidence indicates that Shanghai was the region's financial hub during this period, suggesting that financial development in Hong Kong did not spill over to Shanghai.

¹⁷ We are also note France's defeat in the France-Prussia war in 1870. The defeat may have weakened France's influence and reputation in its colonies, potentially reducing economic and financial development. However, the effect of the France-Prussia war on financial development in French Shanghai seems limited. The number of banks, trading companies, population kept growing in the French Shanghai concession after the France-Prussia war. That is, there is no evidence of a decline in economic and financial development. Furthermore, France's domestic stock market capitalization also did not decline after 1870 (see Figure 4 in La Porta et al, 2008).

these political differences occurring at the empire level could have influenced comparative financial development in the concessions. To account for the possible confounding effect of political institutions, we control for the annual Polity IV democracy scores of Britain and France between 1846 and 1925, where the scores range from -10 (strongly autocratic) to 10 (strongly democratic) (Marshall et al. 2020).

Fifth, we control for Britain's and France's GDP per capita since the relative strength of the British economy, rather than the comparative advantage of the British common law, could shape comparative financial development in the Shanghai concessions. Specifically, we include *GDP per capita*, which equals the logarithm of British/French GDP per capita for the British/French concession respectively. Maddison (2010) provides data on GDP.

Sixth, we extend these robustness tests by controlling for differences in stock market performance in Britain and France. We were concerned that Britain's and France's domestic financial performance could have influenced comparative financial development in their Shanghai concessions. Using data from Campbell et al. (2021), Jordà et al. (2019), and Le Bris and Hautcoeur (2010), we condition on *Stock return*, which equals annual stock returns in London/Paris for the British/French concession respectively.

D.3 Results with additional controls

The results hold when controlling for these additional control variables. Column 1 of Table IV presents the baseline regression that only includes concession and year fixed effects. We include this baseline because, unlike Table II, the data used in Table IV start in 1846 when the control variables become available. Hence, we cannot control for *British*Post1845* (as there is no "pre-treatment" period). Column 2 adds the five concession controls, and Column 3 adds the five empire controls. Across the different specifications, *British*Post1869* enters positively and significantly at the one percent level. After adding the ten controls, the Table IV estimates indicate that the British concession averaged about 9 more banks per year than the French concession after the establishment of the Mixed Courts in 1896.

We were also concerned that the formation of the concessions in 1845 might have triggered different dynamics in the evolution of bank density in the two concessions. Such dynamics could impede the ability to draw sharp inferences about the relationship between legal origins and comparative financial development per se. Thus, we control for linear and quadratic trends that are interacted with the British dummy variable to address this concern (*British*quadratic time trend since 1845*). As shown in Column 4, the results hold when including all of the additional controls and this linear-quadratic trend. Critically, the estimated coefficient on the legal effect, *British*Post1869*, changes little across columns 2-4, suggesting the robustness of the results to the controls. The results also hold when including *Stock return*. Although conditioning on *Stock return* reduces the sample size, Appendix Table AIV demonstrates that *British*Post1869* continues to enter positively and significantly in the financial development regressions.

Finally, we extend the Figure 3 analyses of whether there is a break in the evolution of comparative financial development in the concessions after the formation of the Mixed Courts. Specifically, we augment equation 2 by also controlling for the interaction between each of the control variables in Table IV (fixed in the year 1868) and year dummies. We include these interaction terms to address potential concerns that differences in the concession might create differential trends in comparative financial development. We continue to find a sharp break in. comparative financial development after 1869. However, since most of these individual interaction terms drop from the analyses due to multi-collinearity at the grid-cell level, we report these robustness tests using the first principal component of the Table IV controls in 1868 interacted with year dummies. As shown in Appendix Figure A4, we continue to find a distinct break in the evolution of comparative financial development following the formation of the Mixed Courts, with financial development in the British concession rising appreciably relative to that in the French concession after 1869.

These findings are consistent with the legal origins view of comparative financial development. Namely, they suggest a strong, independent impact of applying the two legal systems on comparative financial development. We do not argue that other factors are unhelpful in explaining comparative financial development in the Shanghai concessions. Rather, the results suggest there was a notable increase in financial development in the British concession relative to that in the French concession with the application of the two distinct legal systems in 1869. We now examine whether there is a similarly notable decrease in comparative financial development in the two concessions with the rendition of those legal systems in 1926.

IV. Comparative Financial Development and Rendition of the Courts

Another legal system break occurred around 1926 with the rendition of the Mixed Courts. Following violent protests in 1925 against Western influence, negotiations intensified about the rendition of the Mixed Courts. The Chinese and the United Kingdom signed an agreement in August of 1926, and the formal rendition of the British Mixed Court to the Chinese authorities occurred on January 1, 1927. Legal jurisdiction in the French concession was also reformed, and Chinese courts took over cases involving Chinese citizens in 1927 and, later, complete legal jurisdiction.

This section tests whether financial development deteriorated comparatively rapidly in the British concession following the rendition. If the British common law provides a comparative advantage over other legal systems in supporting financial development, eliminating this advantage should have an especially large adverse effect on finance in the British concession. We examine twenty years around the rendition year so that our sample covers from 1917 through 1936.

A. Dynamics: Comparative financial development and rendition

To assess whether there was a break in the evolution of comparative financial development around the rendition of the Mixed Courts, we again estimate the dynamic version of equation (1) at the grid cell level with 1926 as the reference year:

$$Y_{it} = \sum_{t=1917}^{1936} \beta_t \times British_i \times \theta_t + \gamma_i + \lambda_t + \varepsilon_{it}, \qquad (3)$$

where θ_t equals one in year t and zero otherwise, and the other variables are defined above. Following the same method used to construct Figure 3, we plot the estimated values of β_t from equation (3) and the corresponding 95% confidence intervals in Figure 4. If the rendition of the Mixed Courts has a disproportionately negative effect on financial development in the British concession, we should see a decline in β_t after 1926.

As shown, there is a discrete drop in the comparative financial development of the British concession following the rendition of the courts. This result holds for the full sample of all banks, as shown in Panel A, and the subsample of Chinese banks, as illustrated in Panel B. The break in the evolution of comparative financial development is consistent with the legal origins view that moving from the British and French legal traditions to the Chinese courts will have a relatively larger adverse effect on financial development in the British concession, given the superiority of the British common law in fostering finance.¹⁸ Following our analyses of the evolution of comparative financial

¹⁸ One potential concern is that the May Thirtieth Movement in 1925 may confound the effect of the court rendition on comparative financial development in the two concessions. However, this nationalistic movement influenced both the British and French concessions in Shanghai and other locales. We have not found empirical evidence or historical analyses indicating that growing Chinese nationalism influenced the British concession more than the French concession. Moreover, our data show that population and public expenditures in the British concession did not fall relative to that in the French concession after 1925. To further address this concern, we restrict the sample period to the period after the end of the Movement (Mid-August 1925) and examine stock price changes around the court rendition event (August 1926).

development around the founding of the Mixed Courts, we also augment these dynamic analyses of the rendition of courts by including additional controls. Specifically, we compute the first principal components of the Table V controls. We then include this first principal component (computed in 1925) interacted with year dummies. Appendix Figure A5 shows that we continue to observe a distinct break in the evolution of comparative financial development following the rendition of the courts, with financial development in the British concession falling relative to that in the French concession after 1926.

B. Rendition and comparative bank density, bank performance, and stock returns

We next employ a difference-in-differences strategy to examine comparative bank density and then extend the analyses to bank performance and stock returns. The first difference compares the two concessions. The second difference involves comparing (1) the period from 1917 to the rendition of those courts in 1926 to (2) the period from the 1926 rendition until 1936, when the Second Sino-Japanese War effectively ended British and French influence in the Shanghai concessions. The legal origins view predicts a more substantial decline in financial development in the British concession than in the French concession following the rendition of the Mixed Courts because removing the British common law will have had a larger adverse impact on financial development than removing the French civil law. We use 1926 as the rendition date for the difference-in-differences estimation even though the official rendition occurred on January 1, 1927, because the agreement was negotiated in 1926 and finalized in August 1926. Using 1927 instead of 1926 as the breakpoint date in the regressions yields almost identical results. We report results for the subset of Chinese banks to save space. Results on all banks are similar, as shown in Appendix Table AV.

Table V shows that the British concession experienced a much sharper decline in Chinese bank density than the French concession after 1926. We again employ the extensive set of controls used in Table IV. Column 1 only includes *British* * *Post1926* and concession and year fixed effects. Column 2 adds the concession controls,¹⁹ column 3 also conditions on the five empire controls, and column 4 further includes linear and quadratic trends interacted with the British dummy variable to address this concern (*British*quadratic time trend since 1845*). Consistent with the legal origins view, *British*Post1926* enters negatively and significantly across the different specifications at least at the 10-percent significance level. Furthermore, the estimated coefficient on *British*Post1926* varies little when using the different conditioning information sets,

¹⁹ *Municipal council* is omitted because both councils were established before the sample period in Table V.

highlighting the strong, independent relationship between legal traditions and comparative financial development. The estimated coefficient from column 4 indicates that the gap in bank density between the British and French concessions fell by about seven banks per km² due to a drop in bank density in the British concession. This is large, as the gap at the start of the sample period (1917) was about 40 per km².²⁰

Next, we extend these analyses to examine bank performance. We collected data on banks' return on assets (ROA) between 1921 and 1931 from a 10-year survey by the Research Department, Bank of China (1933). Although we could only obtain data on 26 banks, we use them to shed additional evidence on the impact of the rendition of the Mixed Courts on comparative bank performance in the concessions. We employ two regression specifications in which the dependent variable is the ROA of a bank in a year. The first specification is a simple linear regression that is separately estimated over (a) the pre-rendition period and (b) the post-rendition period. The primary explanatory variable is *British*, which equals one for banks in the British concession and zero otherwise. The regression controls for year-fixed effects and sometimes conditions on time-varying bank controls. The bank controls include lagged values of the logarithm of bank assets, bank leverage, and the logarithm of bank age. This regression provides information on whether average bank ROAs differ in the two concessions before and after the rendition of the Mixed Courts. For example, if the British common law more effectively supports banking activities, it could lower the banking costs in the British concession. This cost advantage could boost bank ROAs if entry barriers limit the extent to which competition eliminates the cost advantages. Second, we employ a difference-in-differences specification. The main explanatory variable is the interaction term British * Post1926, and the regressions include bank and year-fixed effects and the same time-varying bank controls discussed above. This regression provides information on whether the rendition of the Mixed Courts is associated with differential change bank ROA in the two concessions.

²⁰ There are other potential confounding factors. For instance, Britain's move to the gold standard occurred at about the same time as the rendition of the Mixed Courts. Thus, we cannot include an interaction term of the gold standard and the British concession. However, the analyses condition on many factors that might be associated with adopting the gold standard, including GDP per capita and stock returns in the home countries (Appendix Table AVI), population density in the concessions, and the density of trading companies in the concessions. The estimated coefficient on the *British*Post1927* interaction term changes little when including or excluding these controls, suggesting that omitted variables, e.g., the return to the gold standard, are not significantly biasing the results on the legal origin effect. Another concern is frequent political unrest during this warlord era of China. To address the concession, we control for the interaction between British concession dummy and the annual number of wars. However, we recognize that this control will not eliminate the potential impact of various influences on comparative financial development around 1927. However, we unaware of research showing that such events had more adverse effects on the British concession.

Table VI reports two main findings. First, banks in the British concession had higher ROAs than those in the French concession during the pre-rendition period. As shown in columns (1) and (2), *British* enters positively and significantly during the pre-rendition period when excluding or including the time-varying bank controls. Second, this performance gap disappears following the rendition of the Mixed Courts. As shown in columns (3) and (4), *British* is insignificantly different from zero during the post-rendition period. The difference-in-differences analyses also indicate that the ROA performance gap drops after the rendition of the courts. As shown in columns (3) and (4), *British* * *Post1926* enters negatively and significantly. The findings in Table VI indicate that while bank ROAs were, on average, higher in the British concession before the rendition of the Mixed Courts, they were no longer higher following the rendition of those courts to Chinese rule.

Another way to shed empirical light on the impact of the rendition of the Mixed Courts is to examine the comparative change in the stock returns of firms in the British and French concessions following 1926, which we offer in Table VII. To do this, we examine the 24 months around August 1926, i.e., between September 1925 and August 1927. This short time window excludes the period of the nationalistic movement that ended in August 1925. The dependent variable is the monthly stock return of each company, defined as the log difference of month-end stock prices. The stock return regressions control for time-varying firm characteristics—the logarithm of the firm's market capitalization and the number of months since the firm went public. Several specifications also condition on firm-fixed effects to control for time-invariant traits. To provide a benchmark, we examine the overall 'event effect' on stock returns without distinguishing between the British and French concessions (columns 1 and 2). Column (1) excludes firm fixed effects, while column (2) includes them. As shown, there is a significant drop in stock returns following the rendition of the Mixed Courts, with returns falling by about 3.3% when including the full set of conditioning variables. We then examine the differential stock return response by including the interaction term, British*Post-August 1926, and report the results in columns (3) and (4). In column (3), we exclude the firm fixed effects and include *British* as a separate linear term. We include firm fixed effects in column (4), so British drops from the estimation. The results indicate that stock returns fell by more in the British concession following the rendition of the Mixed Courts. Indeed, the stock returns of the average British firm fell by about 4.6% more than that of the corresponding French firm. The Table VII results provide further evidence that shifting from the British common law to Chinese civil law had a more adverse effect on firms than the concomitant legal system changes in the French concession.

V. Beyond Shanghai

In this section, we examine concessions outside of Shanghai. Shanghai was the financial center of China in the late 19th and early 20th centuries, and there are much better data on the Shanghai concessions than those in other cities. However, examining the non-Shanghai concessions enhances our assessment of how different legal systems within one city shape comparative financial development within those cities.

Specifically, we examine 16 Western concessions in three cities—Tianjin, Hankou, and Guangzhou. The cities and concessions, as well as their legal origins and establishment years, are listed in Appendix Table AVII. We chose these three cities because they hosted concessions with a British common law tradition and concessions with a civil law tradition.²¹ Therefore, we can compare financial development in common and civil law jurisdictions within the same city. To conduct these comparisons, we implement two strategies. First, we divide these concessions into two groups: those with a common law tradition (Britain and the United States) and those with a civil law tradition (Austro-Hungarian Empire, Belgium, France, German, Italy, Japan, and Russia).²² We then test whether financial development is comparatively stronger in the common law concessions. Second, we focus only on the differences between British and French concessions in these three cities. This second strategy directly extends our Shanghai analyses to these three cities and avoids combining different civil and common law concessions. This second strategy comes at the expense of excluding the other (non-British and non-French) concessions. As we will show, both strategies yield the same conclusions.

Given these data, we estimate the following equation over the period between 1850 (ten years before the establishment of the first concession) and 1936 (the eve of the Second Sino-Japanese War):

$$Y_{it} = \beta_1 \times Common \ Law_i \times Post + \beta_2 \times Population \ density_{i-1} + \gamma_i + \lambda_t + \varepsilon_{it}, \quad (4)$$

where Y_{it} denotes the number of banks (per km²) in concession *i*, in year *t*. Common Law_i is a dummy variable that equals one if the concession was granted to a common law country and zero if granted to a civil law country. Post is a dummy variable that

²¹ Ten other cities in China had concessions between circa 1845 and 1945. These other cities, however, had only one concession, so we cannot compare the impacts of applying the common and civil laws on financial development in those cities. Therefore, we exclude them from our analysis. Moreover, the Qing dynasty also ceded Kulangsu of Xiamen to ten Western powers (and Japan) as an international settlement. However, there were no clear concession boundaries within Kulangsu, so we excluded Kulangsu.

²² Internet Appendix Table AVII lists each concession, its city, the foreign country (home country) overseeing the concession, and the year it was established. Internet Appendix B lists data sources.

equals one after the concession was established and zero otherwise. The coefficient on *Common Law_i* × *Post* indicates the estimated average annual difference in bank density between common and civil law concessions. As in the earlier analyses, we control for lagged population density at the concession level (*Population density (lag)*) and concession and year fixed effect. As noted above, we estimate equation (4) for the full sample of concessions (strategy 1) and for the subsample of concessions that includes only British and French concessions (strategy 2).

Table VIII shows that bank density is materially larger in the common law concessions than in their civil law counterparts. When considering all concessions, we find that, on average, the number of banks per km² in the common law concessions is 4.4 greater than that in the civil law concessions (column 2). And this gap is mainly driven by differences in the number of Chinese banks, not differences in the number of foreign banks (column 4). While subject to greater data limitations than the analyses focused on the Shanghai concessions, these findings on 16 concessions outside Shanghai are consistent with those comparing the Shanghai concessions in these cities, as shown in columns (3) and (5). These results provide additional evidence that the common law legal tradition fostered comparatively strong financial development across different concessions within China.

VI. Conclusion

This paper constructs a new database on Western concessions in China from 1845 through 1936 and exploits changes in British and French legal jurisdiction over the respective concessions. We use these data and regime changes to assess the legal origins view of comparative financial development, which stresses that economies with a common law tradition more effectively foster financial development than economies with a civil law tradition. When applied to the Shanghai concessions, the legal origins view predicts that (1) financial development in the British concession should grow comparatively more than that in the French concession after Britain and France established legal jurisdiction over their respective concessions in 1869 and (2) financial development in the British concession should shrink comparatively more after the rendition of the Mixed Courts to Chinese rule in 1926.

The evidence is consistent with the legal origins view of comparative financial development. The financial development advantage of the British concession emerged with the application of the British and French legal systems in the respective concessions in 1869. This financial development gap then shrank after the 1926 rendition of the Mixed Courts that essentially eliminated the application of these

Western legal traditions in the concessions. These results hold when controlling for an array of time-varying military, economic, and political differences between Britain and France and their respective influence within China. Furthermore, when we extend these analyses across 16 concessions in three cities outside of Shanghai, we confirm that financial development is materially greater in the common law concessions than in their civil law counterparts.

References

- Acemoglu, D., Johnson, S., and Robinson, J.A. 2001. The colonial origins of comparative development: An empirical investigation. *American Economic Review*, 91(5): 1369–401.
- Bai, Y., and Kung, J. K. 2015. Diffusing knowledge while spreading God's message: Protestantism and economic prosperity in China, 1840-1920. Journal of the European Economic Association, 13(4), 669–698.
- Barth, J. R., Caprio Jr., G., and Levine, R. 2004. Bank regulation and supervision: what works best? *Journal of Financial intermediation*, 13(2), 205–248.
- Beck, T., Demirgüç-Kunt, A. and Levine, R. 2003a. Law, endowments, and finance. *Journal of Financial Economics*, 70(2), 137–181.
- Beck, T., Demirgüç-Kunt, A., and Levine, R. 2003b. Law and finance: why does legal origin matter? Journal of Comparative Economics, 31(4), 653–675.
- Beck, T., Levine, R., and Loayza, N. 2000. Finance and the sources of growth. *Journal* of Financial Economics, 58(1–2), 261–300.
- Berkowitz, D. and Clay, K., 2011. The Evolution of a Nation: How Geography and Law Shaped the American States. Princeton University Press.
- Berkowitz, D. and Clay, K., 2005. American Civil Law Origins: Implications for State Constitutions and State Courts, American Law and Economics Review 7, 62-84.
- Brown, J. R., Cookson, J. A., and Heimer, R. Z. 2017. Law and finance matter: Lessons from externally imposed courts. *The Review of Financial Studies*, 30(3), 1019– 1051.
- Brown, J. R., Martinsson, G., and Petersen, B. C. 2013. Law, stock markets, and innovation. *The Journal of Finance*, 68(4), 1517–1549.
- Cai, X. 2013. Wanqing Huayang Shangshi Jiufen Yanjiu (On Sino-foreign Commercial Disputes of the Late Qing Dynasty). Beijing: Zhonghua Shuju.
- Chesneau, R. (Ed.). 1979. Conway's All the World's Fighting Ships: 1860-1905. London: Conway Maritime Press.
- Chesneau, R (Ed.). 1980, Conway's All the World's Fighting Ships 1922–1946. Annapolis: Naval Institute Press.
- Chen, X. 1992. Zhongwai Zhanzheng Zhanyi Dacidian (The Dictionary of Chinese Civil and Foreign Wars and Battles). Changsha: Hunan Chuban She.
- Clodfelter, M. 2002. Warfare and Armed Conflicts: A Statistical Reference to Casualty and Other Figures, 1500–2000. Jefferson: McFarland Publishing.
- Colledge, J. J., and Warlow, B. 2010. Ships of the Royal Navy: The Complete Record of all Fighting Ships of the Royal Navy from the 15th Century to the Present. Havertown: Casemate.
- Conseil d'administraction municipal de la concession francaise a Shanghai. 1868-1936. Compte-rendu de la gestion pour l'exercise (1868-1936) Budget (1869-1937). Shanghai: Imprimerie Municipale.

- D'Acunto, F., Prokopczuk, M., and Weber, M. 2019. Historical antisemitism, ethnic specialization, and financial development. *Review of Economic Studies*, 86, 1170–1206.
- D'Acunto, F., From Financial History to History & Finance, Boston College working paper, 2019.
- Demerliac, A. 2021. La Marine de la restauration et de Louis-Philippe 1er: Nomenclature des navires français de 1815 à 1848. Paris: Ancre.
- Djankov, S., Hart, O., McLiesh, C., and Shleifer, A. 2008. Debt enforcement around the world. *Journal of Political Economy*, 116(6), 1105–1149.
- Djankov, S., La Porta, R., Lopez-de-Silanes, F., and Shleifer, A. 2003. Courts. The Quarterly Journal of Economics, 118(2), 453–517.
- Djankov, S., McLiesh, C., and Shleifer, A. 2007. Private credit in 129 countries. *Journal* of financial Economics, 84(2), 299–329.
- Du, X. 2006. Jindai Zhongguo Qianye Xiguan Fa: Yi Shanghai Qianye Wei Shijiao (Customary Law of Traditional Banking in Modern China: Evidence from Shanghai). Shanghai: Shanghai caijing daxue chubanshe.
- Durnev, A., and Kim, E. H. 2005. To steal or not to steal: Firm attributes, legal environment, and valuation. *The Journal of Finance*, 60(3), 1461–1493.
- Fairbank, J.K., 1978. The creation of the treaty system. The Cambridge History of China, 10(part 1), 213–263.
- Fauvel, A. A. 1899. Histoire de la Concession française de Chang-hai. Paris: E. de Soye et Fils, Imprimeur.
- Feuerwerker, A., 1983. The foreign presence in China. The Cambridge History of China, 12(pt 1), 128–208.
- Fisman, R., and Love, I. 2004. Financial development and intersectoral allocation: A new approach. The Journal of Finance, 59(6), 2785–2807.
- Gardiner, R. (Ed.). 1985. Conway's All the World's Fighting Ships: 1906-1921. Annapolis: Naval Institute Press.
- Gennaioli, N., and Shleifer, A. 2007. The evolution of common law. *Journal of Political Economy*, 115(1), 43–68.
- Glaeser, E. L., and Shleifer, A. 2002. Legal origins. The Quarterly Journal of Economics, 117(4), 1193–1229.
- Guiso, L., Sapienza, P., and Zingales, L. 2004. The role of social capital in financial development. *American Economic Review*, 94(3), 526–556.
- Henriot, C., Lu, S., and Aubrun, C. 2018. *The Population of Shanghai (1865-1953): A Sourcebook.* Leiden; Boston: Brill.
- Hou, Q. 2017. Wanqing zhongwai huishen zhidu zhong huayang faguan de falv suyang yu shenpan fengge: Yi Shanghai fazujie huishen gongxie weili (Chinese and foreign judges' trial style and legal literacy in the joint-trial system of late Qing: Evidence from the Mixed Court of the French concession in Shanghai). Xueshu Yuekan, 49(1), 165–176.
- Hudson, M. O. 1927. The rendition of the International Mixed Court at Shanghai. *The* American Journal of International Law, 21(3), 451–471.

- Jappelli, T., Pagano, M., and Bianco, M. 2005. Courts and banks: Effects of judicial Enforcement on Credit Markets. Journal of Money, Credit and Banking, 37(2), 223-244.
- Jernigan, T. R. 1905. China in law and commerce. New York, NY: Macmillan.
- Jia, R. 2014. The legacy of forced freedom: China's treaty ports. *Review of Economics* and Statistics, 96(4), 596–608.
- Keller, W., Santiago, J.A. and Shiue, C.H., 2017. China's domestic trade during the Treaty-Port Era. Explorations in Economic History, 63(1), 26–43.
- Keller, W., and Shiue, C. H. 2021. The economic consequences of the Opium War. Paper presented at NBER Chinese Economy Working Group Meeting, July 2020, Cambridge, MA.
- King, R.G., Levine, R., 1993. Finance and growth: Schumpeter might be right. The Quarterly Journal of Economics, 108(3), 717-737.
- Kung, J. K. 2021. Impact of the West: Findings from new data. In: The Cambridge Economic History of China: Volume 2, 1800 to the Present, Eds: Debin Ma and Richard von Glahn, forthcoming.
- La Porta, R., Lopez-de-Silanes, F., and Shleifer, A. 2006. What works in securities laws? *The Journal of Finance*, 61(1), 1–32.
- La Porta, R., Lopez-de-Silanes, F., and Shleifer, A. 2008. The economic consequences of legal origins. *Journal of Economic Literature*, 46(2), 285–332.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., and Vishny, R. W. 1997. Legal determinants of external finance. *The Journal of Finance*, 52(3), 1131–1150.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., and Vishny, R. W. 1998. Law and finance. Journal of Political Economy, 106(6), 1113–1155.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., and Vishny, R. W. 2002. Investor protection and corporate valuation. *The Journal of Finance*, 57(3), 1147–1170.
- Lee, T.V. 1990. Law and local autonomy at the International Mixed Court of Shanghai. Ph.D. Thesis Yale University.
- Levine, R. 2005. Law, endowments, and property rights. *Journal of Economic Perspectives*, 19(3), 61–88.
- Levine, R., Loayza, N., and Beck, T. 2000. Financial intermediation and growth: Causality and causes. *Journal of Monetary Economics*, 46(1), 31–77.
- Licht, A.N., Goldschmidt, C. and Schwartz, S.H., 2005. Culture, law, and corporate governance. *International Review of Law and Economics*, 25(2), 229-255.
- Liu, Z. 2004. Jindai Shanghai Huashang Zhengquan Shichang Yanjiu (Studies on Chinese Securities Market in Modern Shanghai). Shanghai: Xuelin Chubanshe.
- Maddison, A., 2010. Statistics on world population, GDP, and per capital GDP, 1-2008 AD (Horizontal file), Maddison Historical Statistics. Groningen, the Netherlands: Groningen Growth and Development Centre, University of Groningen.
- Marshall, M. G., Jaggers, K. and Gurr, T.R. 2020. Political regime characteristics and transitions, 1800–2018, Polity IV Project, Center for International Development and Conflict Management at the University of Maryland, College Park.
- Montalto de Jesus, C. A. 1919. *Historic Shanghai*. Shanghai: Shanghai Mercury.

- Morck, R., Wolfenzon, D., and Yeung, B. 2005. Corporate governance, economic entrenchment and growth, *Journal of Economic Literature*, 43, 655–720.
- Musacchio, A., and Turner, J. D. 2013. Does the law and finance hypothesis pass the test of history? *Business History*, 55(4), 524–542.
- Pagano, M., and Volpin, P. F. 2005. The political economy of corporate governance. American Economic Review, 95(4), 1005–1030.
- Pascali, L. 2016. Banks and development: Jewish communities in the Italian Renaissance and current economic performance. *Review of Economics and Statistics*, 98(1), 140-158.
- Perotti, E. C., and Von Thadden, E. L. 2006. The political economy of corporate control and labor rents. *Journal of Political Economy*, 114(1), 145–175.
- Pierce, L., and Snyder, J.A. 2017. The historical slave trade and firm access to finance in Africa. *Review of Financial Studies*, 31(1), 142-174.
- Pistor, K. 2009. Rethinking the "law and finance" paradigm. BYU Law Review, 2009(6), 1647-1670
- Rajan, R. G., and Zingales, L. 2003. The great reversals: the politics of financial development in the twentieth century. *Journal of Financial Economics*, 69(1), 5–50.
- Research Department, Bank of China. 1933. An Analysis of the Accounts of the Principal Chinese Banks, 1921-31. Shanghai: Research Department, Bank of China.
- Roe, M. J. 2006. Legal origins, politics, and modern stock markets. Harvard Law Reiew, 120, 460–527.
- Safavian, M., and Sharma, S. 2007. When do creditor rights work. *Journal of Comparative Economics*, 35(3), 484–508.
- Shanghai Municipal Council. 1863-1937. Report for the Year 1863-1936 and Budget for the Year 1864-1937. Shanghai: Kelly & Walsh Limited.
- Shanghai Shehui Kexueyuan (Shanghai Academy of Social Sciences). 1981. Xinghai Geming Zai Shanghai Shiliao Xuanji (Selected Works on the History of Xinhai Revolution in Shanghai). Shanghai: Renmin Chubanshe.
- Stulz, R. M., and Williamson, R. 2003. Culture, openness, and finance. Journal of Financial Economics, 70(3), 313–349.
- U.S. State Department. 1880. Papers Relating to the Foreign Relations of the United States, Transmitted to Congress, With the Annual Message of the President, December 6, 1880. US Government Printing Office, 1880. Available at https://history.state.gov/historicaldocuments/frus1880/d133
- Voigtländer, N., and Voth, H.-J. 2012. Persecution perpetuated: The medieval origins of anti-semitic violence in Nazi Germany. The Quarterly Journal of Economics, 127(3), 1339-1392.
- Wang J. 1965. Shijiu Shiji Waiguo Qinhua Qiye zhong de Huashang Fugu Huodong (The equity investment by Chinese businessmen in foreign invading enterprises in the 19th century). Historical Research, 4, 39-74.
- Wang, L. 1998. *Shanghai Fazhishi* (Legal History of Shanghai). Shanghai: Renmin Chubanshe.

- Wang, W. 1995. Shanghai Caizheng Shuiwu Zhi (Fiscal History of Shanghai). Shanghai: Shanghai shehui kexueyuan chubanshe.
- Xin, X., and Yan, S. 2021. Zhengzhi Zhidu yu Yinhang Xingwei -- Yi Ershi Shiji Sanshi Niandai Shanghai Zujie Weili(Political System and Banking Behavior: Taking Shanghai Concession in the 1930s as an Example). Lianghua Lishi Yanjiu, 6, 30-70.
- Yan, Z. 1955. Zhongguo jindai jingjishi tongji ziliao xuanji (Selection of Statistical Materials on the Economic History of Modern China). Beijing: Kexue Chuban She.
- Yang, X. 2006. Diguo zhi Bian yu Guatou zhi Lian: Shanghai Huishengongxie Quanli Guanxi Bianqian Yanjiu (The Imperial Whip and the Oligarchic Chain: Study of the Power Transition in the Shanghai Mixed Court). Beijing: Peking University Press.
- Zou, Y. 1980. *Jiu Shanghai Renkou Bianqian de Yanjiu* (Studies on the Population Change of Old Shanghai). Shanghai: Renmin Chubanshe.



Figure 1 Evolution of the Shanghai Concessions

Notes: The figure shows the expansion of the British and French Concessions between 1845 and 1914. The areas in different shades of red depict the British concession at different dates, and those areas highlighted in shades of blue are the French concession. The grey area denotes the Old City of Shanghai County under Chinese administration. The two areas with boundaries outlined with bold, black perimeters, i.e., the dark red and dark blue locations, denote the concessions in 1860, which we call the "old concessions." The old British concession covers 512 acres (2.07km²), and the old French concession covers 180 acres (0.73km²).







Notes: These figures show the distributions of banks at the grid-cell level (200m*200m) within the old concessions. Banks are compared before and after the establishment of the Mixed Courts (1868 and 1925). The British Concession is north of the border, depicted with a bold, black perimeter.



(b) Chinese banks only

Figure 3 Bank Development and the Creation of the Mixed Courts in 1869

Notes: This figure depicts the estimated annual gap in bank density between the British and French concessions before and after the establishment of the Mixed Courts in 1869. Panel (a) uses the number of all banks as the dependent variable, and Panel (b) uses the number of Chinese banks as the dependent variable. We estimate the following regressions at the grid-cell level using the GLS method to address the autocorrelation of residuals:

$$Y_{it} = \sum_{t=1865}^{1880} \beta_t \times British_i \times \theta_t + \gamma_i + \lambda_t + \varepsilon_{it},$$

where Y_{it} is the number of banks within gird-cell *i*, in year *t*, and θ_t equals one in year *t* and zero otherwise, *British* is dummy variable that equals one for grid cells in the British concession and zero otherwise, γ_i and λ_t are grid-cell and year fixed effects respectively, and 1869 is the reference year. The figure plots the estimated values of and confidence intervals around β_t .



(b) Chinese banks only

Figure 4 The Impact of the Court Rendition (1926) on Banking Development

Notes: This figure depicts the estimated annual gap in bank density between the British and French concessions before and after the rendition of the Mixed Courts in 1926. Panel (a) uses the number of all banks as the dependent variable, and Panel (b) uses the number of Chinese banks as the dependent variable. Specifically, we estimate the following regressions at the grid-cell level using the GLS method to address the autocorrelation of residuals. The estimation is at the grid-cell level with 1926 as the reference point:

$$Y_{it} = \sum_{t=1917}^{1936} eta_t imes British_i imes eta_t + \gamma_i + \lambda_t + arepsilon_{it},$$

where Y_{it} is the number of banks within gird-cell *i*, in year *t*, and θ_t equals one in year *t* and zero otherwise, *British* is dummy variable that equals one for grid cells in the British concession and zero otherwise, γ_i and λ_t are grid-cell and year fixed effects respectively, and 1926 is the reference year. The figure plots the estimated values of and confidence intervals around β_t .

Table I

Textual analysis of the verdicts in Shanghai concessions, 1870-1926

This table examines differences in the texts of verdicts in the British and French courts in the Shanghai concessions. The sample includes 406 commercial case verdicts between 1870 and 1926, of which 111 are cases involving banks. For each case, we calculate the ratio of keywords indicated in the first column to the total number of words in the verdict and then multiply by 1,000 to compute the millesimal. We test whether the incidence of these keywords differs between the two Mixed Courts using the indicated t-tests. Internet Appendix Table AI provides details on the keywords. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

	British Mixed Court	French Mixed Court	
	Mean	Mean	Difference
Panel A. All commercial cases	(n=315)	$(n\!\!=\!\!145)$	
words related to custom and convention	0.46	0.18	0.28**
words related to consult and opinion	0.55	0.20	0.35**
words related to default	0.93	0.80	0.13
Panel B. Bank-related cases only	$(n\!\!=\!\!55)$	$(n\!\!=\!\!56)$	
words related to custom and convention	0.61	0.10	0.51***
words related to consult and opinion	0.81	0.24	0.57^{**}
words related to default	1.17	0.69	0.48

Table II

Bank Density and the Formation of the Mixed Courts in the Concessions, 1840-1925

This table presents regression results relating bank density to the formation of the Shanghai Concessions and the formation of the Mixed Courts in the British and French Concessions. In column 1, the dependent variable is the total number of banks per square kilometer in the respective concession in a year. In column 2, the dependent variable is the total number of Chinese-owned banks per square kilometer in the respective concession in a year. In column 3, the dependent variable is the total number of banks per grid cell in a year. In column 4, the dependent variable is the total number of Chineseowned banks per grid cell in a year. We divide the concessions into 200m by 200m grid cells for the gridcell analyses. *British* is a dummy variable that equals one for the British Concession and zero for the French Concession. *Post1845* indicates the formation of the concessions, so it equals one after 1845 and equals zero for the years from 1840 through 1845. *Post1869* indicates the formation of the Mixed Courts, so it equals one after 1869 and equals zero for the years from 1840 through 1869. The regressions control year fixed effects (*Year FE*) and either Concession fixed effects (*Concession FE*) or Grid-cell fixed effects (*Grid-cell FE*) as indicated. We report the GLS estimates that adjust for autocorrelation of the residuals. Standard errors are reported in parentheses and are clustered at the grid-cell level in columns 3 and 4. *, **, and *** indicate significance at the 10%, 5%, and 1% levels, respectively.

	Bank density at	t the concession	Number of banks at the grid-cell			
	lev	vel	level			
	All banks	Chinese banks	All banks	Chinese banks		
	1	2	3	4		
British * Post1845	2.084	1.590	0.082	0.060		
	(2.035)	(1.729)	(0.064)	(0.060)		
British * Post1869	8.147***	6.667***	0.317^{***}	0.265^{***}		
	(1.937)	(1.651)	(0.059)	(0.055)		
Year FE	Yes	Yes	Yes	Yes		
Concession FE	Yes	Yes				
Grid-cell FE			Yes	Yes		
Observations	172	172	8,600	8,600		

Table III

Bank Density and the Formation of the Mixed Courts in the Concessions: Alternative Samples

This table uses the same methodology as in Table II except it (1) restricts the period to the Qing dynasty, 1840-1910 (Panel A), (2) restricts the sample to the shorter and more stable period from 1865 through 1894 (Panel B), and (3) uses data aggregated over five-year periods from year -2 to +2 to reduce the impact of higher frequency influences (Panel C). All columns report the GLS estimates that adjust for autocorrelation of the residuals. Standard errors are reported in parentheses and are clustered at the grid-cell level in columns 3 and 4. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

	Bank density a	at the concession evel	Number of ba cell	nks at the grid- level	
	All banks	Chinese banks	All banks	Chinese banks	
	1	2	3	4	
Panel A		Qing dynast	y, 1840-1910		
British * Post1845	1.885	1.341	0.048	0.031	
	(1.304)	(1.196)	(0.044)	(0.042)	
British * Post1869	8.517***	7.720***	0.222***	0.183^{***}	
	(1.144)	(1.034)	(0.042)	(0.040)	
Year FE	Yes	Yes	Yes	Yes	
Concession FE	Yes	Yes			
Grid-cell FE			Yes	Yes	
Observations	142	142	$7,\!100$	$7,\!100$	
Panel B		Period without	wars,1865-1894		
British * Post1869	4.369***	4.412***	0.126**	0.133**	
	(1.266)	(1.192)	(0.055)	(0.053)	
Year FE	Yes	Yes	Yes	Yes	
Concession FE	Yes	Yes			
Grid-cell FE			Yes	Yes	
Observations	60	60	3,000	3,000	
Panel C		Five-year dat	ia, 1840-1925		
British * Post1845	4.745	3.176	0.194	0.121	
	(6.634)	(5.386)	(0.178)	(0.165)	
British * Post1869	17.207***	14.777***	0.671^{***}	0.541^{***}	
	(5.850)	(4.806)	(0.157)	(0.144)	
Year FE	Yes	Yes	Yes	Yes	
Concession FE	Yes	Yes			
Grid-cell FE			Yes	Yes	
Observations	36	36	$1,\!800$	1,800	

Table IV

Chinese Bank Density and the Formation of the Mixed Courts in the Concessions: Additional Controls, 1846-1925

This table presents regression results relating bank density to the formation of the Mixed Courts in the British and French Concessions while controlling for time-varying characteristics of the concessions and of the homeland countries. The dependent variable is the total number of Chinese banks per square kilometer in the respective concession in a year. Municipal council is a dummy that equals one if a municipal council was established in a concession. The British municipal council was established in 1854, and the French one was established in 1862. *Population density* is defined as the number (in thousands) of total people per square kilometer at the concession level in the previous year. Trade company density is the number of trade companies per square kilometer at the concession level in the previous year. Public expenditure is the annual total spending on public infrastructures and government services (in 1,000 silver Taels) reported by the Municipal Councils in the previous year. Wars in China equals the total number of civil wars and foreign wars in China in the previous year. Concessions in China equals the total number of concessions of Britain or France in China in the previous year. Naval ships denotes the number of naval ships in service in Britain and in France in the previous year. War victories is the number of victories in wars involving Britain or France in the previous year. Polity score ranges from -10 to 10, measuring the level of democracy in Britain and France in the previous year, based on the data of Polity IV. GDP per capita denotes the logarithm of British GDP per capita for the British concession and the logarithm of French GDP per capita for the French concession in the previous year. All columns report GLS estimates that adjust for autocorrelation of the residuals. Standard errors are reported in parentheses. *, **, and *** indicate significance at 10%, 5%, and 1%, respectively.

	Density of Chinese banks				
—	1	2	3	4	
British * Post1869	6.417***	9.373***	9.558***	8.482***	
	(1.714)	(1.929)	(2.637)	(1.832)	
$Concession/China\ characteristics$					
Municipal council		-0.739	1.470	0.394	
		(1.589)	(2.051)	(1.286)	
Population density		0.033	0.020	0.063**	
		(0.039)	(0.048)	(0.030)	
Public expenditure		0.257	0.563	0.181	
		(0.330)	(0.460)	(0.280)	
Trade company density		0.340***	0.573^{***}	0.124	
		(0.114)	(0.146)	(0.094)	
British * Wars in China		-0.209	-0.468**	0.293^{*}	
		(0.182)	(0.228)	(0.150)	
British/French colonial power					
Concessions in China			1.301^{**}	2.298^{***}	
			(0.600)	(0.586)	
Naval ships			0.018^{***}	-0.002	
			(0.002)	(0.002)	
War victories			0.336	0.474^{***}	
			(0.276)	(0.171)	
Polity score			-0.179	-0.155	
			(0.200)	(0.124)	
GDP per capita			-32.592^{***}	0.252	
			(4.803)	(3.508)	
Year and Concession FEs	Yes	Yes	Yes	Yes	
British * Quadratic time trend since 1845				Yes	
Observations	160	160	160	160	

Table V

Chinese Bank Density and the Rendition of the Mixed Courts in the Concessions in 1926

This table examines the comparative changes in Chinese bank density between the British and French Concessions after the agreement regarding the rendition of the Mixed Courts in 1926. The sample covers the period from 1917 through 1937, i.e., twenty years around the rendition year. The control variables are identical to those of Table IV, except that *Municipal council* is omitted because both councils were established before the sample period in the respective concession. All columns report the GLS estimates that adjust for autocorrelation of the residuals. *, **, and *** indicate significance at 10%, 5% and 1%, respectively.

	Density of Chinese banks					
-	1	2	3	4		
British * Post1926	-5.021*	-7.346*	-6.533***	-7.326***		
	(2.674)	(3.761)	(2.131)	(1.713)		
Concession/China characteristics						
Population density		1.193^{***}	0.857***	0.621^{***}		
		(0.320)	(0.251)	(0.193)		
Public expenditure		-2.346	-16.327***	-15.056^{***}		
		(6.200)	(3.567)	(2.683)		
Trade company density		2.605***	2.997***	1.681^{***}		
		(0.385)	(0.357)	(0.580)		
British * Wars in China		0.812	0.853	0.586		
		(0.655)	(0.549)	(0.527)		
British/French colonial power						
Concessions in China			1.839***	-0.239		
			(0.647)	(0.625)		
Naval ships			0.002	-0.003		
-			(0.004)	(0.003)		
War victories			-1.904***	-0.470		
			(0.401)	(0.380)		
Polity score			0.560	0.448		
-			(1.530)	(1.203)		
GDP per capita			-2.055	4.743		
			(8.763)	(7.299)		
Year and Concession FEs	Yes	Yes	Yes	Yes		
British * Quadratic time trend since				Yes		
1917						
Observations	40	40	40	40		

Table VI

Bank Performance and the Rendition of the Mixed Courts

This table examines comparative bank performance in the concessions after the rendition of the Mixed Courts to Chinese jurisdiction in 1926. The dependent variable is the Return on Asset (ROA, *100), i.e., net profit divided by asset, at the bank level. *Bank controls* include the logarithm of assets in the previous year, the leverage (loan to asset ratio, *100) in the previous year, and the logarithm of bank age. The sample consists of 26 major banks between 1921 and 1931. Standard errors are clustered at the firm level and reported in parentheses. *, **, and *** indicate significance at 10%, 5% and 1%, respectively.

	Pre-Rendition period 1921-1926		Post-Rendition period 1927-1931		Difference-in- Differences estimation, 1921-1931		
	1	2	3	4	5	6	
British	0.819**	1.136^{***}	-0.194	-0.253			
	(0.363)	(0.393)	(0.401)	(0.460)			
British * Post1926					-1.106**	-1.296*	
					(0.525)	(0.669)	
Bank controls		Yes		Yes		Yes	
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	
Bank FE					Yes	Yes	
R-squared	0.138	0.527	0.028	0.056	0.420	0.393	
Observations	138	112	123	119	261	231	

Table VII Stock Returns and the Rendition of the Mixed Courts

This table provides regression results relating stock returns to the rendition of the Mixed Courts to Chinese jurisdiction. The dependent variable is a firm's monthly stock return, defined as the log difference of month-end prices, i.e., $\log(\text{price}_t) - \log(\text{price}_{t-1})$. The sample period is from September 1925 through August of 1927, i.e., a 24-month window around the month of the court rendition announcement (August 1926). Post-August 1926 is a dummy that equals one for the months after the August 1926 rendition announcement. *Firm controls* include the log duration (of months) since the firm went public (listed) and the log market capitalization. *Firm FE* denotes the firm fixed effects that absorb all time-invariant firm characteristics. Standard errors are clustered at the firm level and reported in parentheses. *, **, and *** indicate significance at 10%, 5% and 1%, respectively.

	Stock return					
_	1	2	3	4		
Post-August 1926	-0.039***	-0.033***	0.004	0.011		
	(0.008)	(0.010)	(0.006)	(0.009)		
British * Post-August 1926			-0.045***	-0.046***		
			(0.010)	(0.011)		
British			0.029^{***}			
			(0.008)			
Firm controls	Yes	Yes	Yes	Yes		
Firm FE		Yes		Yes		
R-squared	0.011	0.081	0.012	0.082		
Observations	1,404	1,404	1,404	1,404		

Table VIII

Legal Origins and Finance: Concessions in other Chinese Cities, 1850-1936

This table provides regression results relating bank density to the legal origin of the foreign entity with legal jurisdiction over 16 concessions in three cities (Tianjin, Hankou, and Guangzhou). Appendix Table AVII provides details on the countries and legal traditions with jurisdiction over these concessions. In columns 1-3, the dependent variable is the total number of banks per square kilometer in the respective concession in a year. In columns 4 and 5, the dependent variable is the total number of Chinese-owned banks per square kilometer in the respective concession in a year. In columns 4 and 5, the dependent variable is the total number of Chinese-owned banks per square kilometer in the respective concession in a year. Common Law is a dummy variable that equals one for the concessions ruled by common law countries and zero for the concessions ruled by civil law countries. Post is a dummy variable that equals one after the concession was established. Population density is the number (in thousands) of total people per square kilometer at the concession-year level. In columns 3 and 5, we restrict the analysis to only the concessions of Britain and France to compare the British common law to French civil law. The regressions control for year-fixed effects (Year FE) and concession fixed effects (Concession FE) as indicated- Heteroskedasticity robust standard errors are reported in parentheses. *, **, and *** indicate significance at the 10%, 5% and 1% levels, respectively.

		All banks	Chinese banks		
			British vs.		British vs.
	All	All	French	All	French
	concessions	concessions	concessions	concessions	$\operatorname{concessions}$
	1	2	3	4	5
Common Law * Post	5.911***	4.970***	4.511***	4.384***	3.793***
	(1.091)	(1.231)	(1.342)	(1.139)	(1.271)
Population density (lag)		0.135^{***}	0.225^{***}	0.134^{***}	0.227^{***}
		(0.027)	(0.029)	(0.028)	(0.030)
Year FE	Yes	Yes	Yes	Yes	Yes
Concession FE	Yes	Yes	Yes	Yes	Yes
R-Squared	0.354	0.440	0.635	0.436	0.629
Observations	$1,\!392$	$1,\!392$	522	1,392	522

Internet Appendix

A. Figures and Tables



Figure A1. The Mixed Court at the British Concession of Shanghai (1869 to 1926)

Source: Wikipedia,

 $https://commons.wikimedia.org/wiki/File:International_Mixed_Court_at_Shangha~i7.jpg$

International States Share Paid Latest Div International States Share Share Paid Latest Div International States Share Share Share Share Share International States Share	June 9th, 1866.	The Rorth-China Daily Detos.		1761					
Article Jack Hall is Article Banks. Blaired Paid Istor Direction Banks. Banks. Blaired Paid Istor E <	Abe Bieft-china Bill Ber-		MACKENZIE & CO.	PUBLIC AUCTION.	SHARES AN	D 870	CER		
Banks. Shared Paid Latest Dr up. Price. dan Commercial Bank Corporation. Comptoir d'Eacompte de Part of India, Comptoir d'Eacompte de Part of India, Comptoir d'Eacompte de Part of India, Arga & Masternan's Bank, of India, London and Chattered Bank of Hindustan Chattered Bank of Hindustan Chattered Bank of Hindustan Chattered Bank of Hindustan Chattered Marcantile Bank of India, London and Chattered Marcantile	Surrenz, Jose Son, 1999.	a proble, predice a fait definitioned interfacional description of the problem of the problem of the definition of the problem	AT SALES ROOM,	Jarvie Therburn & Co.	SHANES AN	D 810	сьз.		1
Banks. Shared Value Drives don up. Price. don	c) Anny, (seem that (see a state of the s	An intervent Company in quilted, additionally meanser Company in quilted, additionally addates for for regions. The menoighal is addete the regions. The intervent is a second of the region of addates for for regions of the means of the regions. The intervent is a second of the region of the addates of the means of the means of the region of the regions of the region of the region of the addates of the means of the means of the region of the intervent of the means of the region of the reg	 Burris Educe Paint Paint Const Back Paint Street Unit Statistics Conductionery, Linearized Conductionery, Linearized Vision 	A start of the sta	1922 18	1	Data	Taturt	Diat
X Up. PTRON Construction Image: State of the state	Hongborg on the fifth Fast, and have on the State of a strain into part on the 15th Fast, a per Artikan.	Stant Nergislan Company offy mere lange have dely per supplier in the block competition in the property off of the per vector of the block of the supplier of the block of the block of the supplier of the supplier of the block of the supplier of the block of the supplier of the block of the supplier of the supplier of the supplier of the supplier of the block of the supplier of the supplier of the block of the supplier of the supplice of the supplier of	· Date Fing Tuberson · Date and Copp. Bit. 10. An	 State Westward - Andreas for Space and Be developed Designing for any real by Frank Indian. State State State State State State State State 	Banks.	Shares	Faid	Delest	DIVI
Within the second se	The Family Hall of May 1755, data in Hong W long as the Diff. Ann, and have on the 201 of the Adds. B control have pair on the lat of Addy	there has able of format reason is an opto full Shory of mins. Some . I represe a reversely is the point of the deputies of the mean is the weak of the matched format of the ma	10 I-las Practis Busine. 14 Autor Lasing Kim-ma. And	Statistic province, it would be foregated that have been up to be the second seco	100002604020		up.	Price.	dend.
Oriental Bank Corporation. 2 2 4 2 Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." Iohang." I	per Mo. Ashear-Investor New publication of the North China Datip Zows of pagesement of the Don a R.	et in funct. All this longery water proof of fully in the property, and the protective is the other proof of the longer at the fully of the longer at the second se	R Same Sping Dec.	All their valuation property with the Weldinger thermon, histories on the dynamic Menning Allisted in the control of the Allisted Antonion and pro-	The set of				
Oriental Bank Corporation. 25 25 20 Comptoir d'Escompte de Paris 25 25 20 Paris Sank of India, 20 10 15 Charterel Bank of India, 20 20 20 20 Charterel Mercantile Bank of India, 20 20 20 20 Charterel Mercantile Bank of India, 20 20 20 20 The Bank of India, London and China, 20 10 15 25 15 Bank of India, London and China, 20 100 15 20 20 20 Comptoir d'Esconpation, 3125 127 16	AND DESCRIPTION OF TAXABLE ST	ing, peptidates are able will will be able and many higher are block, and higher and higher and higher are able and higher	Se 10 Breght bir Jan, 165.	To which it is bounded on the New Yorking South it is builded by the provide comparing the former sector and the former sec- tor sector and the sector and the sector and the					
Contential Bank Corporation. 25 25 25 20 Commercial Bank of India, Agra & Masterman's Bank Comptoir d'Escompte de Paris Comptoir d'Escompte de Paris Contrered Bank of India, Agra & Masterman's Bank Contrered Bank of India, Banking Company Shanglai Club Shanglai S. N. Co Show 1.000 1.000 North China insurance Co Show 31 S. N. Co Show 31 S. N. Co Show 50 800 Nil. North China insurance	"Rang," "Filler II." "For thes," "Init's Social, "Shark," and "Eriptic", "Lad yas," manifers, "Space," of Social. Environment in Social Soc	definitely local provide local and the second secon	ALL TRATAGE	The second		*	£	£	£
Commercial Bank Corpora- tion Comptoir d'Escompte de Paris Astate Bank of India, Astate Bank of India, Astate Bank of India, Astate Bank of India, Dartered Bank of Hindustan Chartered Bank of Hindustan China and Japan Commercial Bank of Hindustan China and Japan Commercial Cos. The Bank of Hindustan China and Japan Commercial Sciences Commercial Sciences Commercial Sciences Commercial Sciences Commercial Bank of Hindustan China and Japan Commercial Cos. Shanghai Dock Co. Shanghai Dock Co. Shan	Ba the "Res of China" Fick Ind 1aft Loss for the set of	control in the Simulate One Cone is much agreements of an attach time of the Simulation of the Simu	The 1955 Junear,	If your 7 has an elements. The includes, investor an the above property, common all Produces of the probability theoremetricities	Oriental Bank Corporation.	25	25	47	
the t	in Minard, 41 her a canonaire of gebra? Too the 12 R.A. semileday in manie in moment the days po Press the train of Polescore to the Equation, to Lange B114 W. representation for standard, to 14	represented, segment and designed where the second	AT MT SALES BOOK,	Diving Room, Denning Room, and Room Red. Sector or the primary Room, and Four Red. Sector while Denning Rooms, and Red. Sciences	Commercial Bank Corpora-	8			
Comptoir d'Escompte de Paris de la service d	change to X of the root of the data of the last winds and the worker. Buy the proof the Ar best show the sectord of Monoray , taken the part of the merger light variable wind with re-	remains the gluon which president in the format and shales is not and a freque to prove of press. In the format is the structure of the struct	THE RECORD LARGE T	The One lowers conserved with the Bradling Intere comprehension of Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro Alexandro A	tion	25	25	20	S
Paris	Winners bill dep ren. Jose 6 5 in Lat. 95.01 10 8. Lange 183 21 H. spiler the order "data"	o collar, The French Gan Company, in the first data and the first interficient environment grand and the providing providing in 27 states with the state of the first interficient and the state of the state of the state of the state of the state of t	Cohan c."	There is a linear teacher and lines in feel- and a teach concerney in an entertain of the investing binnes. Interact, any statement of the first	Comptoir d'Escompte de				
Bank of India, 50 25 51 8 p. Astatic Bank of India, 20 20 20 20 Astatic Bank of India, 20 20 20 20 20 Astatic Bank of India, 20	Bank by "Persite" Ho which In In In Repairwell 1941 which side with of the R and of Research, where a drag get from 3	Which is morphism has also in hereiner, in the second particular set or an end particular set on the second particular set of the se	vooung,	Investigation of the proving, and memory first data up 10 K status data we meansions after which is second at a story estimating titles and a final first house an information.	Paris	8 an 6	1 100000	1000	
Agra & Masterman's Bank 50 25 51 50 Agra & Masterman's Bank 20 10 15 Australia and China 20 20 20 Australia and China 20 20 20 The Bank of Hindustan China and Japan 100 25 25 15 The Bank of Hindustan China and Japan 100 25 125 127 19 Shanglai Dock Co. 500 500 50 7 <td< td=""><td>and by which indeed for likene does it ballied parts of of the parage had fight ratio is which with an ratio and associated fight. Awited here parker of dat. (2)</td><td>anappend. To expediate on the view by part for our manual sites of the part of the strength of</td><td>The multi-scienced has have inclusioned in net by PEDRAF CATCRY as non-values DAVID for Membry, the 1-to response</td><td>Events and Judica and a senior ress. Events Events and Franking Research Relieve Reside. Water Learner and the side Learner Park Relieve</td><td>Bank of India</td><td></td><td></td><td>••</td><td>•••</td></td<>	and by which indeed for likene does it ballied parts of of the parage had fight ratio is which with an ratio and associated fight. Awited here parker of dat. (2)	anappend. To expediate on the view by part for our manual sites of the part of the strength of	The multi-scienced has have inclusioned in net by PEDRAF CATCRY as non-values DAVID for Membry, the 1-to response	Events and Judica and a senior ress. Events Events and Franking Research Relieve Reside. Water Learner and the side Learner Park Relieve	Bank of India			••	•••
Astatic Banking Corporation. 20 20 20 20 Astatic Bank of India, Australia and China. 20 20 20 20 Astatic Bank of Hindustan China and Japan 100 25 25 43 134p Image: State of Bank of Hindustan China and Japan 100 25 15 Image: State of Bank of Hindustan China and Japan 100 25 15 Image: State of Bank of Hindustan China and Japan 100 25 15 Image: State of Bank of Hindustan China and Japan 100 25 15 Image: State of Bank of Hindustan China and Japan 100 200 200 200 200 Image: State of Bank of Hindustan China and China 100 25 15 15 Image: State of Bank of Hindustan China and China 100 100 15 15 Image: State of Bank of Hindustan China and China Case Oo 1000 1000 100 1000 10 Image: State of Bank of Hindustan China file Instrance Oo 1000	Ann. In part "Non Ward," "France," on it is in the second state of	Spirst Company and the Kangkes interface Manage result is to have a set of the set of	private contrast, the well-known Brilling, in complete result for all and on the private of the more for all moders of the	Berner an under der Bestehnen einer Berner der Bertetten Bertet	Agra & Masterman's Bank				0
Absolute characteristic Bank of India, 20 20 Chartered Bank of India, 20 20 Chartered Bank of Industan Chartered Bank of Industan	which continued for farm fars. Letter per the of prompt light X.E. which and fire dust has maken. Redwol here parameter.	inter different entationneme in hand bestehrungen die ober einer eine	1 University of Reason Party A. Co., King que Boards. The remarkation discontight fixed Code: and (Dep in: and, pr. serve to her many in de- ter many in the serve to her many in de- ter many in the serve to her many in de- ter many in the serve to her many in de- ter many in the serve to her many in de- ter many in the serve to her many in de- ter many in the serve to her many in ther many	1. He Facility of three two Solitons, (data works) as an of data and for embryon bang all of threads, is well extend to for an day and the information of the soliton of	Asiatia Usabinatia Balin	00	20	51	8 p. c.
Chartered Bank of India. 20 20 Chartered Bank of India. Chartered Bank of Hindustan Chartered Bank of Hindustan China and Japan Iou 25 15 Hongkoug and Shanghai Banking Company \$125 125 125 127 94 p. Commercial Cos. Shanghai Dock Co. Shanghai Cab Shanghai Cab Shanghai Cab Shanghai Cab Shanghai Cab Shanghai Cab Shanghai Bank of Line Shanghai Dock Co. Shanghai Cab Shanghai Cab Shanghai Bank of Line Shanghai Dock Co. Shanghai Cab Shanghai Cab Shanghai Bank of Line Shanghai Dock Co. Shanghai Cab Shanghai Bank of Line Shanghai Dock Co. Shanghai Cab Shanghai Bank of Line Shanghai Bank of Co. Shanghai Bank of Jimak Shanghai Bank of Jimak Sh	Ranter, Famil - Responsed of Functions Co.	Impage well by this time have also. Such as the property and program in the property of the second well as the property of the second well as the	ANY EXCEPTION FOR THE ANY EXCEPTION FOR THE ANY EXCEPTION FOR THE ANY EXCEPTION FOR THE ANY EXCEPTION ANY EXCEPTIO	T minings. Browner for the laws is A second circle associating 100 facility 80 facts and the laws of the laws of the single of the sec	Asiatic Dauking Corporation.	20	10	15	
Autor and a constraint of the second of t	bear up the re of theirs, by a sub- ection in the the per university of the Data day pend on "Array" with here in finance of the pend on the sub-	the second secon	The sale is made to observe partnership and mill positivity he will reach the model of more server.	Andreas and a second se	Chartered Bank of India,	1100			
Image: State of the state	al Storagine. Single Children and State States -	ht we fair that the reg 2 presided there much more to explore using a flag hand to be red by the fair of the rest	The INTESTIGATION on his work of an office where poster are defined and in the first transition	 [1] [1] A An, with A relia host, Yee Sense and F Hed Roman with Societaneous climited control stress in terms. 	Australia and China	20	20	20	
Image: State of the state		I the descriptions. Orthogonal of its for dimension have been dear for an intervent with the orthogonal or and the orthogonal orthog	Terms of Hole -then third of the par- sion money to se paid on fail of the horse our. The holescore in Terms range of the	The role locus management is likeling 40 8 4216 1 Auto, and Auto that and the Tim Marcus and Tang to contractions and Will articulate - Microsov for lar and the two find samplinged	Chartered Mercantile Bank	a 8			
The Bank of Hindustan 25 25 43 13 μp The Bank of Hindustan 100 25 15 The Bank of Hindustan 100 25 125 125 127 The Bank of Hindustan 100 100 100 15 The Bank of Hindustan 100 100 100 15 The Bank of Hindustan 100 100 15 The Bank of Hindustan 100 100 100 100 The Bank of Hindustan 100 100 100	which cais for senarit, is that which all ducine Mr. Church's intention of senal-	the instead is an period of the plance of th	the depart is to firstine). Thi expose a transfer in the larger in the per-teasy. N. Markallin.	a) Underson, door is often a revertified within an Inc started to study a sector. A pay on the back sub-of the transmer law described, there is a a shift would get the start is because mean.	of India, London and				
Answer	assumers, that we are quite reliably in- formed the reaco for Re. Church's ar- not a few days sizes, was the discovery of	ander af dar ontrol, that instruments: or probably is downted in this frame, the former of the transmission or the former of the transmission or the former of the transmission of the tra	Trija Kial Monglei, Nie Jani, 1 str.	 The decaying process are explored using exclusion have fore foregap, and proceeding a prime surveyor from Alongue Hall. The enablings over 6 the rest induction and enablings over 6 the rest. 	China	25	25	43	1810 0
Image: Constrained in the second	of his intention to have by the form he all all all all all all all all all al	ene rempfiel juit un invin them for the function of the funct	MONDAY,	as present are increased in terms in the line of a first sector of the line	The Bank of Hindustan	- T			1080.0
 Нопgkoug and Shanghai Shanghai Cas Co. Shonghai Shonghai Shonghai Shonghai Shonghai Shonghai Shonghai Shonghai Cas Co. Shonghai Cas Co. Shonghai Cas Co. Shonghai Shonghai Shong	We reprot to have the total how of the "th during over infere with near back on her way down from Fankers. For way	ind, to found teach.	AT 11.4.16	3 long, and only existing the balance property of the second seco	China and Japan	100	05	10	
Image: State of the state	heng tored arises by the Cherger, it when the elected on to a rash which the stands the fait shows the water or	Concerning the entropy of the second se	SALID: BOOM Deterior and presupery take-folds fol	man in Junge Am June, and animating 10 men in a fin has. CEDERADA.	Honskoug and Shanghai	100		10	••
 Полнания сопрону	mediandly. Unly a potion of the sieve -	ne e mane wa strange wa strange in her	per banded, size en per banded, size en per banded, size en per banded, size en	and both of the book has been been and and should be been been been been been been been	Hanking C. Bhanghan	+10r	100	10-	
При в тока При в тока </td <td>We have renaized a most interesting</td> <td>All legist up led possible dimensi dene [Lee war de hair to des tels dats, being deg gift, wang tes fitte-states prive]] is a fin dene ser tes base to des tels dats, being deg gift, wang tes fitte-states prive]] is a fit de set and tes de ser tes des tels de set and tes de</td> <td>W - Champer Wine, W - "Champer Les Kon" Charge 10 - Reporter First Wine, Fill & co.</td> <td>Lat 10, measures received Ability spaces that, There is an account present that its present that, The population or each late, and had not all of the populations.</td> <td>Danking Company</td> <td>\$125</td> <td>125</td> <td>12.1</td> <td>91 p.c.</td>	We have renaized a most interesting	All legist up led possible dimensi dene [Lee war de hair to des tels dats, being deg gift, wang tes fitte-states prive]] is a fin dene ser tes base to des tels dats, being deg gift, wang tes fitte-states prive]] is a fit de set and tes de ser tes des tels de set and tes de	W - Champer Wine, W - "Champer Les Kon" Charge 10 - Reporter First Wine, Fill & co.	Lat 10, measures received Ability spaces that, There is an account present that its present that, The population or each late, and had not all of the populations.	Danking Company	\$125	125	12.1	91 p.c.
Полнание Полнание Полнание Полнание Полнание Полнание Полнание Пол	report of the program of the Policy in Biophal, drawn up by Dr. Dalgors, pro- make whose care the solution and in	and is the finite pointing proving. The party parameter of the personne on the string are particly dataset, foring as the	No - 104 Tom Gin. 10 - Annual Marra, Sprace, and 10 - Champer Oder, Diodust	Lat 3.4. In estuard or far frank, and Lat 11, a shark pressure that frankque, as bits atted be limit due that.		-			
COMMERCIAL Cos. Tis. Tis. 500 500 500 500 500 500 500 500	man rather repairs frians, or mept-	International and a state of the state	 Cano. Cano. Compagin Broady. Bo Blab Dath State and Toxic Water. Int Canor a Urb. Barr. 	Decad in Lod Americania Barrier Barrier Distantia and company in the Grand Americania and and the State Company					
The second se	terming description of and comments as on Chinese contons. The extent to which as the practice of infasticide percults in to	and the sport implement of all of a first adds there are an importangiand in the sport implement of a sport implement of a sport implement of a sport of a sport implement of a sport of a	50 _ Juffrey's Ince. Abo. 13 Come Diagon Tolanov	which was 11 A. Foldward, 1 and a pair (10 for dash, An e. R. samaling within a for and P. and R. and R. and R. and An e. R. samaling within a formation of the samaling on the Neurok and Association (Section States on the States).	COMMERCIAL COS.	1	1	1	
Imagina in the second secon	discussed at length, and Dr. Dulgens in roman to the conclusion that it is much loss common than is represented. The	series and series on a	10 Vigna Tala.es. Na. 19 Ebb. Online Ecte Plan.	The first of the property of the second are the first of the first of the second region of an observe on par With (theo), No. 16, no analogo is not we done to first	and the second	Tls.	Tls.	Tla.	
A result of the second seco	conveniences of forth, are in his opinion as an Brisset to account for the majority of an dance of independent, without reacting	and all a space of a second of a finite second seco	N	Public Hours, and on the first and West by Chinese event properties to an interact properties where on their investment of the Bernie Car-	Shanghai Dock Co.	500	500	200	None
An analysis	to the surgician of wills nucleor; a white many shiften whom bother are found append, may probably have don't	The second	Alan, 50 Game Rept. Lost regar, 8 and 10 lb.	(bod, Job. 100). Buildwall of angle & Kan, W.S. & Kana, Suman and Suman a	Shanghai Club	450	45.0	15	
Horsgane Wharf Co. 1,000 1,000 16 Horsgane Wharf Co. 1,000 1,000 1000 Horsgane Wharf Co. 1,000 1,000 1000 Horsgane Wharf Co. 1,000 1,000 1000 Horsgane Wharf Co. 1,000 1,000 1,000 Horsgane Wharf Co. 1,000 1,000 1,000 Horsgane Wharf Co. 1,000 1,000 1,000 Horsgane Wharf Co. 5,000 500 800 Horsgane Wharf Co. 5,000 500 1000 North China Insurance Co. 5,000 500 90 Stangeni Brick Co. 1000 1000 1000 Stangeni Brick Co. 500 500 500 Stangeni Brick Co. 500 500 50 Stangeni Brick Co. 500 50 52 30 Temper Temper </td <td>been pervented from bringing their fi- shifteen to the dead-brane by melality of to use the spectrum exected by the of</td> <td>Sended of Marce difficit and addings Before C, Tr., Gaussers, Gau, Wall, C. W., Gausser, Gau</td> <td>Aller, 100 Cores Officear's Manes, armyololog sha to office Electric (proof armed)</td> <td>These datas has a facility as a set of standard and their particity to the strend cannot columns their scheme for a statement point of view, and mandre their may eligible forestantiat.</td> <td>Shan diai (ine (in</td> <td>160</td> <td>100</td> <td>-6</td> <td>,,</td>	been pervented from bringing their fi- shifteen to the dead-brane by melality of to use the spectrum exected by the of	Sended of Marce difficit and addings Before C, Tr., Gaussers, Gau, Wall, C. W., Gausser, Gau	Aller, 100 Cores Officear's Manes, armyololog sha to office Electric (proof armed)	These datas has a facility as a set of standard and their particity to the strend cannot columns their scheme for a statement point of view, and mandre their may eligible forestantiat.	Shan diai (ine (in	160	100	-6	,,
Image: Strange and Stra	Boats in charge. We have not space in to review the report fully in this more in ing's paper; two shall do so at greater th	with the of Samady, we description for the second secon	He case of the or. Contine. St D one Paint roll. Alan.	ECCELANS. The W Local Assoc (From in March, 1992) of white of Local Assoc (From in March, 1992) and	Howard Whart Che	1 000	1 000	10	,,
Image: Standard W gart Co	Ir has been adjusted to us that, in the	named a mean of search and a performance of the second and a second a	Not invite Far's Special Early 18 min 1 day, such front Youngers, 18 min 2 day, such front Youngers,	er bei, Jun, 24 m. Ster. Fillelich Consultation formalish om före Jungel for sind an effektive for an en	and an	1,000	1,000	10	"
Shang'a I S. N. Go. 1,000 600 , Image: A state of the state of t	incriming a few days age the different in losing executations to be note in the or course of an afternam's stord forcegit U	and bidewards provide in the second second second bideward in the second bideward in the second seco	Finds and in good strict.	26. E. lageber with its residue buildings second theorem. Constitution of the second second a vary constraints. Dealing Home with Officer, the locate, Congregative Manage, 2 Days	Shaughat What I Co	1,000	1,000		.,,
Image: Second	the orthogen, we don't the interference of the section of the size within the mittle.	In Proceedings of a provide sector of the control o	Friday & San and Provide Contracts. 19 hits main & large mpt. General Budget. Alon. In Section of the University of the	many, and a map forborn, the wheth integra- bes was been the integration of an integrate deviation. The entire property incommunity by a first. This and the manual group of an appro-	Shanghal S. N. Co.	1,000	1,000	600	"
North China insurance Co 5,000 1,000 Strick & Saw Mill Co S50 850 25 Shanghai Brick Co S00 500 Nill Shanghai Brick Co S00 500 850 25 Shanghai Brick Co S00 500 500	tants. We now endearour to reactly in these eremights, by giving a list of the second state of the second	Needs Vision Takin menungkala in and a spectar and applied for pro- Needs Vision Takin menungkala in and a spectar and applied for pro- Needs Vision Takin menungkala in and he spectar and applied for pro- transmed 4 brok menungkala in and he spectar and applied for pro- transmed 4 brok menungkala in and he spectar and applied for pro- transmed 4 brok menungkala in and he spectar and applied for pro- transmed 4 brok menungkala in and he spectar and he spectar and applied for pro- transmed 4 brok menungkala in and he spectar and he spe	A Case way Says Legran thes just to Plane Saint Dranks Tanada (Incide Wear Flanet)	Laster for the second s	China rire Insurance Co	5,000	<u><u></u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u>	800	
S Brick & Saw Mill Co	and the value of their shares, in a wi- similar form to the list of shares and a stasks published daily in the Xoglish in	to be descriptioned them thanks legal to be the second state of the line line waves to be a second state of the second state o	1 Cherry Dentes, 21 day, 1 Cherry Long Room, 21 day, 1 and 20 and 20 and 10 Miles.	2 Million and a test of Local domand use the distribution Concentration, in the strends pressing probabilities, and insurationity behind the filled, implement on the distribution Constrainty on the Acad Sci, and	North China Insurance Co	5,000	1.000		100
S Tu & C Lighter Co S50 850 25 ,, Shanginai Brick Co S50 50 0.75 ,, Sixugata Horas thatas the second	a collating the information on ford, and doubt not that definencies and in-	we refer to the set of the total has been standing to the set of the set of the total has been standing to the set of	Note and ded Morphic flow, 20 years	Introduction of the second sec	S. Brick & Saw Mill Co.	500	500	Nil.	None
*Shanghai Brick Co 50 50 0.75 ;; French Gas Co \$50 50 52 ;; *In course of formation.	minuth-up will be midded with the Me remote of inserekiheled. Shares anding to exclusive To. 1.000 reduced to a value by	and of the majored concepts of the indication addam of the transmission of the transmission of the transmission of the transmission of the heart of the degree of the transmission of th	Alin, 2 arts Double Encome.	This property, well known as the S-Siless Hong, was not holds tables the candid empirica- tendence of an Kanapara Artikian. The etheration of such are given it to available	S. They & Lighter Co	\$50	850	95	
Image genes, 50<	of To. 100, and others brought down al- adenity to a spplar are the rule rule- than the energies; and with the sole	responses of the second	Tijn 1100 Biorgini, Just Bor, 1915.	Describes in Backness, is include to the property inter-show another. It is non-stray induced and its revery way admitted a singular for business products.	1 *Shavehai Brick Co-	100	000	~	,,
French Gas Co	and Sorth Chins Instrument Comparine, day and Sorth Chins Instrument Comparine, day pot a sant of dividend was designed has	and a function of the subscription of the one of the start of the star	PERSONAL PROPERTY AND A PROPERTY AND	A first of family in the Bas Stary risk, sample- big first that in drych by the start bar is handed, between the same start by one Tangana Sinon, and "strated there is the Same. Tang MS2, transform	Shan man Harris Co	100			"
Predict Gas Co	additional subject for compute block is an in-	end dari Palay, but hay need substrate a set of the proceedings of a set of the set of t	Made II. Irad	with the tablings complete	Balaughar Horse Elazaar,	50	50	0.75	
* In course of formation.	and the value of such share will be all of remain a few additional units by the its simple from of comparison. The oper test	hade for everal degr had non-polet an in- ter of the data and the second	The basic by scar of the U.S. One	of still party, both still the Chinese Estimate Reving a forward of the Rev. and estimating bothwell Hill fam. Unk Assess and a logarith of the Statement Hill fam.	French Gas Co	\$50	, 50	52	,,
	minut of the North China Lourance of Company appear to have been open, ally successful. Their principle of	Annual formation formation Annual formation of the second	de Anorises repair d'une al Pr tere Espisie, orde cor lie of Tang da day. The read and in imposed a sep time	The Table to this property is a Chinese Table. For exploring the data for the chinese transmission, we the table and of the origin table. When much of the origin table.	and an and the second sec	1			6.95
	an is smither that designation of a younty in a bit of the second state of the second	a add right of that failed the start of the second start of the second start of the second start balance and start of the second start of the seco	And a state of the	paties, and in the second for the particular at its the second population graphic in any in dimensional on application to P. Karthalas,	* In course of farmer	ion			
		the second s		t a new a rear manglei, né spri, 1985		1011.	1. C		

Figure A2. Record of Stock Quotation in Shanghai Exchange in 1866

Source: The North-China Herald, section "Shares and Stocks" on June 9th, 1866.



Figure A3. Population Density in 1935

Notes: This figure depicts the population density by census district in Shanghai concessions in 1935, based on Henriot et al. (2018).



Figure A4. Bank Development and the Creation of the Mixed Courts in 1869: Additional Controls

This figure replicates Figure 3 (Panel b) but additionally controls for the first principal component of the controls in column (3) of Table IV measured in 1868 interacted with year dummies.



Figure A5. The Impact of the Court Rendition (1926) on Banking Development: Additional Controls

This figure replicates Figure 4 (Panel b) but additionally controls for the first principal component of the controls in column (3) of Table V measured in 1925 interacted with year dummies.

Variable	Obs.	Mean	S.D.	Min	Max
Concession-year level					
Bank density	194	17.8	25.1	0	95.7
Chinese bank density	194	14.8	21.5	0	82.6
Municipal council	182	0.9	0.4	0	1
Population density (1,000	182	9 0 0	10.7	4.6	100 C
$habitants/km^2$)		38.8	19.7	4.0	100.0
Public expenditure (1,000 Taels)	182	1,240.4	2,862.7	0	$16,\!312.6$
Trade company density	182	8.3	7.3	0	25.4
Wars in China	182	3.4	4.1	0	15
Concessions in China	182	4.9	3.3	0	10
Naval ships	182	406.0	428.0	7	2218
War victories	182	1.8	1.5	0	6
Polity IV score	182	5.3	4.8	-8	10
GDP per capita (in Int. GK\$)	182	$3,\!298.4$	$1,\!138.1$	$1,\!478.3$	6,035.2
Grid-year level					
Number of banks	9,700	0.6	2.6	0	47
Number of Chinese banks	9,700	0.5	2.4	0	47
Bank-year level					
Return on Asset (ROA, $*100$)	261	2.1	2.3	-23.8	9.9
Firm-month level					
Stock return (Sept 1925-Aug 1927)	$1,\!626$	-0.004	0.21	-2.48	3.40
Concessions in other cities					
Bank density	$1,\!392$	1.8	8.7	0	85.4
Chinese bank density	$1,\!392$	1.7	8.6	0	85.4
Population density $(1,000 \text{ habitants/km}^2)$	1,392	13.3	27.3	0	246.5

Table AI. Summary Statistics

Category	Keywords
Custom and	English: custom, convention
Convention	English, consult, confor, original ricer
Consult and opinion	English: consult, confer, opinion, view French: consulter, opinion, avis
Default	English: bankrupt, default, unpaid, infringement, breach, non- fulfilment, violation Erench: faillite_non-paiment_violation

Table AII. Keywords for textual analysis of verdicts

Table AIII. Bank Density and the Formation of the Mixed Courts in the Concessions: Additional Controls, 1846-1925

This table replicates Table IV except that the dependent variable is the total number of banks (both Chinese and foreign banks) per square kilometer in the respective concession in a year. All columns report GLS estimates after adjusting the autocorrelation of residuals. Standard errors are reported in parentheses and are clustered at the grid-cell level in columns 3 and 4. *, **, and *** indicate significance at 10%, 5% and 1%, respectively.

	Density of all banks				
—	1	2	3	4	
British * Post1869	7.817***	11.591***	10.255^{***}	9.802***	
	(2.010)	(2.294)	(3.049)	(2.255)	
Concession/China characteristics					
Municipal council		-0.829	1.728	0.743	
		(1.916)	(2.360)	(1.580)	
Population density		0.027	0.024	0.066*	
		(0.047)	(0.056)	(0.037)	
Public expenditure		0.367	0.675	0.277	
		(0.398)	(0.526)	(0.344)	
Trade company density		0.449^{***}	0.709^{***}	0.209^{*}	
		(0.137)	(0.168)	(0.116)	
British * Wars in China		-0.230	-0.465*	0.358^{*}	
		(0.219)	(0.263)	(0.184)	
British/French colonial power					
Concessions in China			1.841***	3.033^{***}	
			(0.697)	(0.721)	
Naval ships			0.021^{***}	-0.000	
			(0.002)	(0.002)	
War victories			0.362	0.545^{***}	
			(0.318)	(0.210)	
Polity score			-0.205	-0.175	
			(0.230)	(0.152)	
GDP per capita			-36.304***	-1.370	
			(5.520)	(4.300)	
Year and Concession FEs	Yes	Yes	Yes	Yes	
British * Quadratic time trend since 1845				Yes	
Observations	160	160	160	160	

Table AIV. Chinese Bank Density and the Formation of the MixedCourts: Additional Controls, 1855-1925

This table augments the analyses of columns (3) and (4) in Table IV by adding *Stock return*, which equals annual stock returns in London/Paris for the British/French concession respectively. The data are from Campbell et al. (2021), Jordà et al. (2019), and Le Bris and Hautcoeur (2010). Due to the data limits, the sample period in this table is from 1855 to 1925. All columns report the GLS estimates after adjusting the autocorrelation of residuals. *, **, and *** indicate significance at 10%, 5% and 1%, respectively.

	Density of Chinese banks		
	1	2	
British * Post1869	7.216**	4.672***	
	(3.459)	(1.801)	
Concession/China characteristics			
Municipal council	-0.010	0.018	
	(0.055)	(0.029)	
Population density	2.541	-4.462***	
	(3.028)	(1.577)	
Public expenditure	0.598	0.106	
	(0.495)	(0.249)	
Trade company density	0.676***	0.162^{*}	
	(0.170)	(0.090)	
British * Wars in China	-0.529*	-0.187	
	(0.286)	(0.165)	
British/French colonial power			
Concessions in China	0.666	1.759^{***}	
	(0.974)	(0.559)	
Naval ships	0.018***	-0.003**	
	(0.002)	(0.002)	
War victories	0.339	0.331^{**}	
	(0.310)	(0.162)	
Polity score	-0.601	-0.979***	
	(0.398)	(0.213)	
GDP per capita	-35.156***	2.086	
	(5.345)	(3.283)	
Stock return	-2.786	-1.484	
	(3.623)	(1.796)	
Year and Concession FEs	Yes	Yes	
British * Quadratic time trend since 1845		Yes	
Observations	142	142	

Table AV. Bank Density and the Rendition of the Mixed Courts in 1926

This table replicates Table V except that the dependent variable is the total number of banks (both Chinese and foreign banks) per square kilometer in the respective concession in a year. All columns report the GLS estimates after adjusting the autocorrelation of residuals. *, **, and *** indicate significance at 10%, 5% and 1%, respectively.

	Density of all banks			
-	1	2	3	$\frac{4}{4}$
British * Post1926	-4.232	-7.739*	-7.101***	-9.640***
	(3.118)	(4.138)	(2.223)	(1.863)
Concession/China characteristics				
Population density		1.422***	1.013^{***}	0.646***
		(0.351)	(0.268)	(0.210)
Public expenditure		-2.064	-19.157***	-16.103***
		(6.845)	(3.718)	(2.901)
Trade companies density		3.292***	4.038***	1.844***
		(0.415)	(0.394)	(0.644)
British * Wars in China		1.078	1.352**	0.505
		(0.737)	(0.627)	(0.593)
British/French colonial power				
Concessions in China			2.381^{***}	0.013
			(0.731)	(0.685)
Naval ships			0.002	-0.001
			(0.004)	(0.003)
War victories			-1.740***	-0.375
			(0.427)	(0.408)
Polity score			0.218	0.515
			(1.746)	(1.329)
GDP per capita			2.668	3.409
			(9.842)	(8.049)
Year and Concession FEs	Yes	Yes	Yes	Yes
British * Quadratic time trend since				V
1917				Yes
Observations	40	40	40	40

Table AVI. Chinese Bank Density and the Rendition of the Mixed Courts in the Concessions in 1926

This table augments the analyses of columns (3) and (4) in Table V by adding *Stock return*, which equals annual stock returns in London/Paris for the British/French concession respectively. The data are from Campbell et al. (2021), Jordà et al. (2019), and Le Bris and Hautcoeur (2010). The sample covers the period from 1917 through 1937, i.e., twenty years around the rendition year. All columns report the GLS estimates after adjusting the autocorrelation of residuals. *, **, and *** indicate significance at 10%, 5% and 1%, respectively.

	Density of Chinese banks		
	1	2	
British * Post1926	-6.062***	-5.801***	
	(2.207)	(1.471)	
Concession/China characteristics			
Population density	0.783***	0.429**	
	(0.254)	(0.171)	
Public expenditure	-16.868***	-15.742***	
	(3.609)	(2.221)	
Trade company density	2.863***	1.063**	
	(0.363)	(0.531)	
British * Wars in China	0.705	0.233	
	(0.537)	(0.485)	
British/French colonial power			
Concessions in China	2.088***	-0.109	
	(0.666)	(0.528)	
Naval ships	0.003	-0.001	
	(0.004)	(0.003)	
War victories	-1.784***	0.237	
	(0.410)	(0.357)	
Polity score	0.949	1.520	
	(1.502)	(1.075)	
GDP per capita	-6.179	-3.879	
	(8.934)	(6.764)	
Stock return	3.753	10.056***	
	(3.269)	(2.497)	
Year and Concession FEs	Yes	Yes	
British * Quadratic time trend since 1917		Yes	
Observations	40	40	

			Legal	Year of
City	Concession	Home Country	origin	establishment
Tianjin	British concession in Tientsin	Britain	Common	1860
	American concession in Tientsin	United States	Common	1860
	French concession in Tianjin	France	Civil	1861
	(Concession française de Tientsin)			
	German concession in Tianjin	Germany	Civil	1895
	Japanese concession in Tianjin	Japan	Civil	1898
	Russian concession in Tianjin	Russia	Civil	1900
	Italian concession in Tianjin (La	Italy	Civil	1902
	Concessione italiana di Tientsin)			
	Austro-Hungarian concession in	Austro-Hungary	Civil	1902
	Tianjin			
	Belgian concession in Tianjin	Belgium	Civil	1902
Hankou	British Concession in Hankou	Britain	Common	1861
	French Concession in Hankou	France	Civil	1896
	Japanese concession in Hankou	Japan	Civil	1898
	Russian concession in Hankou	Russia	Civil	1896
	German concession in Hankou	Germany	Civil	1895
Guangzhou	British concession in Shamian Island	Britain	Common	1861
	French Concession in Shamian Island	France	Civil	1861

Table AVII. Other Concessions in China between 1860 and 1936

Appendix B. Data Sources

1. Data sources for banks

- Bank of Communications. 1995. Zhongguo Jiaotong Yinhang Shiliao, Diyijuan: 1907– 1949 (Historical Materials on the Chinese Bank of Communications, Volume One: 1907–1949). Beijing: Zhongguo Jinrong Chubanshe.
- Bu, M. 1995. *Zhongguo Yinhang Hangshi, 1912–1949* (The History of the Bank of China, 1912–1949). Beijing: Zhongguo Jinrong Chubanshe.
- Chen, X., Gu, T., and Wang, W. 2000. *Zhongguo Tongshang Yinhang* (The Imperial Bank of China). Shanghai: Shanghai Renmin Chubanshe.
- Ge, Y. 2009. *Huyou Zaji* (Shanghai Travel Miscellanea). Shanghai: Shanghai Shudian Chubanshe.
- Jiang, H. 1990. Zhongguo Difang Yinhangshi (A History of Local Banks in China). Changsha: Hunan Chubanshe.
- Jiang, J. 2014. Jindai Zhongguo Yinhangye Jigou Renming Dacidian (Dictionary of Modern Chinese Banking Institutions and Personages). Shanghai: Shanghai Guji Chubanshe.
- Li, Q., and Zhao, W. 1978. *Shanghai Zhi Qianzhuang* (The Money Houses in Shanghai). Taipei: Huashi Chubanshe.
- Lin, Z. 1928. *Shanghai Shangye Minglu* (Shanghai Business Directory). Shanghai: The Commercial Press.
- North China Daily News. 1872-1941. The North-China Desk Hong List (1872-1941). Shanghai: North China Daily News.
- Quanguo Tushuguan Wenxian Suowei Fuzhi Zhongxin. 2010. Minguo Shiqi Shanghai Shiliao Wenxian Congbian, Diyice (Historical Materials of Shanghai in the Republican Period, First Volume). Beijing: Quanguo Tushuguan Wenxian Suowei Fuzhi Zhongxin.
- Shanghai Tongshe. 1984. Shanghai Yanjiu Ziliao Xubian (Continuation of Shanghai Study Materials). Shanghai: Shanghai Shudian Chubanshe.
- Shangwu Yinshuguan Bianyisuo. 1922, 1925, 1930, and 1936. *Shanghai Zhinan* (Shanghai Directory). Shanghai: The Commercial Press.
- The People's Bank of China Shanghai Branch. 1960. Shanghai Qianzhuang Shiliao (Historical Materials on Money Houses in Shanghai). Shanghai: Shanghai Renmin Chubanshe.
- Wang, Z. 1934. Zhongguo zhi Chuxu Yinhang Shi (A History of Savings Banks in China). Beijing: Zhishi Chanquan Chubanshe.
- Xu, K. 1920. Shanghai Shangye Minglu (Shanghai Business Directory). Shanghai: The Commercial Press.
- Zhongguo Yinhang Jingji Yanjiushi. 1934-1937. *Quanguo Yinhang Nianjian* (National Banking Yearbook). Beijing: Zhongguo Yinhang Jingji Yanjiushi.

2. Data source for British/French stock return

- Campbell, G., Grossman, R. S., and Turner, J. D. 2021. Before the cult of equity: the British stock market, 1829–1929. European Review of Economic History, 25(4), 645-679.
- Jordà, Ò., Knoll, K., Kuvshinov, D., Schularick, M., and Taylor, A. M. 2019. The rate of return on everything, 1870–2015. The Quarterly Journal of Economics, 134(3), 1225-1298.
- Le Bris, D., and Hautcoeur, P. C. 2010. A challenge to triumphant optimists? A blue chips index for the Paris stock exchange, 1854–2007. *Financial History Review*, 17(2), 141-183.

3. Data source for population in other concessions of China

- Chen, W., and Lai, X. 2004. *Tianjin de Renkou Bianqian* (Population Change in Tianjin). Tianjin: Tianjin Guji Chubanshe.
- Fei, C. 1991. *Zhongguo Zujie Shi* (Concessions in Historical China). Shanghai: Shanghai Shehui Kexueyuan Chubanshe.
- Guangzhou Difangzhi Bianzuan Weiyuanhui Bangongshi. 1995. Jindai Guangzhou Kou-an Jingji Shehui Gaikuang (Social-Economic Overview of Modern Guangzhou Port). Jinan: Jinan Daxye Chubanshe.
- Huang, X. 1990. Zhenjiang Wenshi Ziliao (Literature and History Materials of Zhenjiang). Beijing: Zhongguo Wenshi Chubanshe.
- Jin, P., and He, Y. 1989. Hangzhou Gongchenqiao Rizujie de Jige Wenti (Issues on the Japanese Concession in Hangzhou Gongchenqiao), Hangzhou Daxue Xuebao, Zhexue Shehui Kexue Ban, 4, 144–153.
- Jin, P., and He, Y. 1992. Hangzhou Gongchenqiao Rizujie dui Hangzhou de Yingxiang (The Impact of Gongchenqiao Japanese Concession on Hangzhou). Hangzhou Daxue Xuebao, Zhexue Shehui Kexue Ban, 1, 110–117.
- Li, G., and Lin, W. 2006. *Qingdai Guangzhou Shisanhang Jilue* (A Brief Account of the Thirteen *Hongs* of Guangzhou in the Qing Dynasty). Guangzhou: Guangdong Renmin Chubanshe.
- Li, J. 1990. *Tianjin Renkou Shi* (Population History of Tianjin). Tianjin: Nankai Daxue Chubanshe.
- Osato, H., and Son, A. 2011. *Zujie Yanjiu Xin Dongtai* (Perspective of the Concession Studies). Shanghai: Shanghai Renmin Chubanshe.
- Tianjin Dang-an Guan. 1992. *Tianjin Zujie Dang-an Xuanbian* (Selected Archives of Tianjin Concession). Tianjin: Tianjin Renmin Chubanshe.
- Tianjinshi Difangzhi Bianxiu Weiyuanhui. 2001. *Tianjin Tongzhi* (Gazetteer of Tianjin). Tianjin: Tianjin Renmin Chubanshe.
- Yuan, J. 2003. *Hankou Zujie Zhi* (History of Hankou Concession). Wuhan: Wuhan Chubanshe.
- Zhang, H. 1993. *Tianjin de Renkou Bianqian* (Population Changes in Tianjin). Tianjin: Tianjin Renmin Chubanshe.

- Zhengxie Wuhu Xian Wenshi Weiyuanhui. 1994. *Wuhu Xian Wenshi Ziliao, Di Si Ji* (Literature and History Materials of Wuhu Xian, Volume 4). Beijing: Zhongguo Wenshi Chubanshe.
- Zhongguo Renmin Zhengzhi Xieshang Huiyi Guangzhoushi Weiyuanhui Wenshi Ziliao Yanjiu Weiyuanhui. 1991. *Guangzhou Wenshi Ziliao* (Literature and History Materials of Guangzhou). Guangzhou: Guangdong Renmin Chubanshe.
- Zhou, D. 2009. *Hankou de Zujie: Yixiang Lishi Shehuixue de Kaocha* (The Concession of Hankou: A Survey Based on Historical Sociology). Tianjin: Tianjin Jiaoyu Chubanshe.